UNIVERSITY OF EDUCATION, WINNEBA SCHOOL OF BUSINESS

SUSTAINABLE PROCUREMENT ADOPTION AND SUSTAINABLE PERFORMANCE: THE ROLE OF ISOMORPHIC PRESSURES



MASTER OF BUSINESS ADMINISTRATION

UNIVERSITY OF EDUCATION, WINNEBA

SUSTAINABLE PROCUREMENT ADOPTION AND SUSTAINABLE PERFORMANCE: THE ROLE OF ISOMORPHIC PRESSURES



A dissertation submitted to the Department of Procurement and Supply Chain Management, School of Business, in partial fulfilment of the requirements for the award of degree of Master of Business Administration (Procurement and Supply Chain Management) in the university of education, winneba

NOVEMBER, 2023

DECLARATION

Candidates Declaration

I hereby declare that this thesis is the result of my original research and that no part of it has been presented for another degree in this university of elsewhere.

Candidate: Afful Nyarkoh Joyce

Signature.....

Date.....

Supervisors' Declaration



We hereby declare that the preparation and presentation of the thesis were supervised in accordance with the guidelines on supervision of thesis laid down by the University of Education, Winneba.

Supervisor: Ms. Mavis Agbodza

Signature.....

Date.....

DEDICATION

I dedicate this study to Almighty God for all His protection and guidance. Also, to my family especially Mr. Moses Adomako and Madam Hannah Otemah for their emotional and financial support in realizing this beautiful dream.



ACKNOWLEDGEMENTS

First, I owe it to the Almighty God who has seen me through this course, His name be praised. A very big appreciation goes to my Supervisor, Miss Mavis Agbodza for her coaching, committed support in nurturing my dream and constructive suggestions, guidance, professional advice and valuable time devoted throughout this research. It was through her direction that made this project a success.

A big thank you goes to my family whose encouragement and unyielding support made my education successful. I also extend my gratitude to all lecturers of the Procurement and Supply Chain Department who taught us in this programme thus enriching my research with in-depth knowledge I say ayekoo to all of you. Finally, I acknowledge all authors whose work were consulted in the course of this study.



TABLE OF CONTENTS

Content	Page
DECLARATION	iii
DEDICATION	iv
ACKNOWLEDGEMENTS	V
TABLE OF CONTENTS	vi
LIST OF TABLES	ix
LIST OF FIGURES	Х
LIST OF ACRONYMS	xi
ABSTRACT	xii
CHAPTER ONE: INTRODUCTION	1
1.0 Background to the Study	1
1.1 Statement of the Problem	4
1.2 Purpose of the Study	6
1.3 Research Objectives	6
1.4 Research Questions	7
1.5 Significance of the Study	7
1.6 Delimitations	8
1.7 Limitations	8
1.8 Definition of Terms	9
1.9 Organisation of the Study	10
CHAPTER TWO: LITRATURE REVIEW	11
2.0 Introduction	11
2.1.0 Conceptual Review	11
2.2 Empirical Review	17

2.3 Theoretical Review and Hypothesis Development	25
2.4 Hypothesis Development	29
2.5 Conceptual Framework	32
2.6 Chapter Summary	33
CHAPTER THREE: RESEARCH METHODS	35
3.0 Introduction	35
3.1 Research Philosophy	35
3.2 Research Approach	36
3.3 Research Design	37
3.4 Study Area	38
3.5 Study Population	38
3.6 Sample and Sampling Procedure	39
3.7 Data Collection Instruments	40
3.8 Ethical Consideration	41
3.9 Operationalisation of Key Variables	41
3.10 Data collection Procedure	42
3.11 Data Processing and Analysis	43
3.12 Chapter Summary	46
CHAPTER FOUR: RESULTS AND DISCUSSION	47
4.0 Introduction	47
4.1 Response Rate	47
4.2 Demographic Characteristics of Respondents	47
4.3 Measurement Model Assessment	50
4.4 Reflective measurement Model Evaluation	51
4.5 Indicator Factor Loadings	51

4.6 Reliability and Convergent Validity	53
4.7 Discriminant Validity	55
4.5 Assessment of Structural Model	56
4.6 Multicollinearity Statistics	56
4.7 R-Squared	57
4.8 F-Square	58
4.9 Hypothesis Testing	58
4.10 Discussion of findings	61
4.11 Chapter Summary	67

CHAPTER FIVE: SUMMARY, CONCLUSIONS AND

RECOMMENDATIONS	68
5.0 Introduction	68
5.1 Summary	68
5.2 Summary of Key Findings	69
5.3 Conclusion	70
5.4 Recommendations	70
5.5 Contribution to Knowledge	72
5.6 Suggestions for Future Studies	72
REFERENCES	74
APPENDIX	92

LIST OF TABLES

Table		Page
1	Measurement of Variables and Sources	42
2	Demographic Characteristics of Respondents	48
3	Reliability and convergent validity	54
4	HTMT Correlation	56
5	Collinearity statistics	57
6	R-Squared	57
7	F-square	58
8	Hypothesis testing	59



LIST OF FIGURES

Figure	Page
1: Conceptual Framework	33
2: Indicator loadings	52
3:T-Values	52



LIST OF ACRONYMS

SPPA Sustainable Public Procurement Adoption COE **Coercive Pressures** MIM Mimetic Pressures NORM Normative Pressures SP Sustainable Performance PPA Public Procurement Authority Environmental Protection Agency EPA Sustainable Development Goals SDG

ABSTRACT

The study examined the relationship between isomorphic pressures, sustainable procurement adoption and performance of public entities in Ghana. The study used an explanatory research design and a quantitative research approach. Relying on institutional and stakeholder theories, data from 307 public entities were gathered using structured questionnaires and analysed with frequencies, means, standard deviation, partial least squares structural equation modelling (PLS-SEM). Simple random sampling technique was used to determine the sample size. The study proved that isomorphic (coercive, normative and mimetic) pressures influenced sustainable public procurement adoption. Similarly, sustainable procurement adoption mediated the relationship between isomorphic pressures and sustainable performance. The study recommended that institutional regulatory bodies like the Public Procurement Authority of Ghana should spearhead the sustainable operations of public procurement. The findings of the study proved that isomorphic pressures (normative, coercive and mimetic pressures) have a positive and significant effect on sustainable procurement adoption in Ghana. All the three hypotheses were supported. The second objective sought to examine the effect of sustainable procurement adoption on sustainable performance in Ghana. One hypothesis was formulated and tested at the end of the analysis, proving that sustainable public procurement adoption significantly influences sustainable performance. The third objective assessed the mediating role of sustainable procurement adoption in the relationship between isomorphic pressures (coercive, normative, mimetic) and sustainable performance.



CHAPTER ONE

INTRODUCTION

1.0 Background to the Study

The Sustainable Development Goals (SDGs), which span environmental, social, and economic development components, are currently seen as a way to ensure the continuation of the world's development objectives beyond the Millennium Development Goals (MDGs) (Adjei-Bamfo,2019). This has significantly increased scholars' and practitioners' concentration on investigating the topic of sustainability (Mani et al., 2020; Yong et al., 2020). To accomplish SDGs, it is vital to implement sustainable practices throughout the entire scope of corporate activity (Marujo & Casais, 2021). Since adopting the United Nations (UN) SDGs in 2015, Members have tried to enhance Sustainable Public Procurement (SPP) under SDG Goal 12 and Target 7.

SPP incorporates environmental, social, and economic issues in procuring goods, works, and services by government agencies or public institutions (Walker et al., 2008; Brammer & Walker, 2011). Government acquires public goods, works and services for national development (Public Procurement Authority, 2013; World Bank, 2015). The global pursuit for sustainable consumption and production (SDG 12) has put public entities now under the scrutiny of a wide range of industry stakeholders who are aware of the sectors' continuous impact on carbon footprints, climate change, resources and energy use as well as inequality issues when it comes contract awards (Harland et al., 2021). These advancements happen as governments procure goods and contract works and services from domestic and foreign enterprises to benefit their citizens (Baah et al., 2021).

Public procurement is a substantial government activity that financial records nearly 15-20% globally, and national revenue of 20-70% in developing economies (Adjei, 2010; Adjei-Bamfo et al., 2019), indicating that public procurement is gradually recognized as a tool of government policy and a conduit for broader economic, social, and environmental changes (Raj et al., 2020; Edman, 2021; Fuertes Giné, 2022). Thus, public procurement institutions are positioned to contribute to the sustainable use of resources in all sectors (Acquah et al., 2021; Baah et al., 2021). With SGD goals in mind, suppliers, customers, competitors, associations, trade unions, the media, other social entities, and corporations are keenly interested in sustainable adoption and implementation by public entities worldwide (Rob et al., 2019).

Public procurement institutions behaviour in acquiring goods, contracts of works, and service consultations must be improved and redesigned to consider sustainability objectives rather than the traditional procurement method (Adjei-Bamfo et al., 2019). Since the traditional function of procurement has been to emphasize price and simply economic advantages, environmental and social considerations must be merged into the procurement process (Dza et al., 2019; Raj et al., 2020; Acquah et al., 2021). Institutional isomorphism theory was employed to explain changes in organizational behaviour in terms of practice and structure. Businesses and organisations are susceptible to pressures from external forces like governments, customers, suppliers, trade unions, professional bodies, and associations. The stakeholder theory (Freeman et al. 2010; Freeman, 2015; Gray,1995) compliments the institutional theory in this study on the need to meet stakeholders' expectations through pressures exerted on public entities. An entity's performance is determined not only by meeting the expectations of its shareholders but also by stakeholders who may be affected or affect the firm's operations (Lokuwaduge & Heenetigala, 2017).

Institutional isomorphism is defined in this research as institutional pressures that cause changes in a firm's internal behaviours toward convergence to sustainable procurement implementation and adoption by public procurement bodies (Hsu et al., 2013). Despite the significance of SPP promotion in sustainable development, scholars such as Perez-Batres et al. (2011); Cao et al. (2014); Raj et al. (2020); Acquah et al. (2021) have identified mimetic isomorphism as one in which firm conform to a new practice imposed by an external authority through adopting the legitimate practices of leading firms, and coercive isomorphism as firm conformance to a new practice imposed by an external authority, due to the power the external authority wills. Normative isomorphism arises from firms' desire to adhere to the standards, norms, and values of permitted systems and processes established by appropriate professional associations.

Integrating sustainable mechanisms into procurement has the added benefit of improving the ethical behaviour of public entities, invariably improving the same, for suppliers and contractors in particular as well as the general public, by improving air and water quality, reducing harmful emissions and waste generation; increasing society's wealth and health (thus raising standards of living); improving working conditions health and safety, responsible consumption, and labour standards; and decreasing labour agitation (Sethi, Martell & Demir, 2017). Consolidating the achievement of SDG 1, 3, 8, 10, 12, and 13 through Ghana's public procurement, organisations operate through actors like staff and managers to produce the expected outcome (Meyer & Rowan, 1977).

Almost all of the challenges in implementing SPP in Ghana are related to the internal capabilities of public entities (Adusei & Awunyo-Vito, 2015), such as knowledge and skills on sustainability (Asare & Prempeh, 2016), financial constraints

(Grandia, 2016), resistance to change (Brammer & Walker, 2011; Grandia, 2016), civil servant commitment (Adjei-Bamfo, 2019), omission of organizational targets (Gidigah et al., 2021), and lack of procurement professionalism (Kyeremeh, 2022) constrains the effort to adhere to SPP policies. Hence, there is a need for institutional isomorphic pressure on sustainable public procurement adoption. In public procurement, political, environmental, social, and citizen pressure toward sustainability is significant in adopting new rules and practices (Raj, Agrahari, & Srivastava, 2020). The study examines the role of isomorphic pressure on sustainable procurement adoption and performance of public entities in Ghana via the lens of institutional and stakeholder theory.

1.1 Statement of the Problem

The global public procurement market is approximately \$ 13 trillion annually (World Bank, 2016). Developed economies account for more than \$ 12 trillion while developing economies account for approximately \$ 1 trillion annually. Ghana spent \$5,330,916,384 on public procurement, approximately 9% of the country's overall GDP (Auditor General report, 2021). The cost-intensity nature of public procurement can be a lever to encourage sustainability and a circular economy in the procurement of goods, works and services in Ghana. It is against this driver of SDG Goals and circular economy that the Government of Ghana amended its sustainable public procurement process of Ghana. With the dynamics in modern procurement practices, there is a need to manage the various risks that arise from environmental and social factors.

Despite support and reinforcement from governments, procurement agencies, and authorities, the importance of SPP in promoting SDG is yet to become part of established practices (Ibrahim et al., 2017; Adjei-Bamfo, 2019; Harland et al., 2019;

Kyeremeh, 2022). Evidence of the consequences of public procurement to this inaction has been witnessed and felt in various government corporations and public entities. For example, Between 2010 and 2017, the government of Ghana lost over GHS 12 billion (\$2.96 billion in nominal terms) in procurement contracts on sole or restrictive tendering (Danquah Institute, 2016). The government could have saved an estimated \$1.93 billion if a more competitive approach encouraging SME participation and local entrepreneurship was used to select suppliers (Hatch, 2017).

On social issues, insight from Open Contracting Scoping Study Report (2017) indicated an estimated value of about GHS 298 million spent by the government on international tendering against GHS 24 million in national competitive tendering in public procurement contracts. Out of the \$11 trillion spent annually on public procurement, only 1% is awarded to women-owned businesses (World bank, 2021). In Ghana, only 30% of contracts are mandatorily allocated to women, youth, and persons with disability-based firms, which is relatively low for the three categories of marginalised firms. Again, in 2021, Ghana's procurement irregularities amounted to a cost of GHS306,769,261.45, and these irregularities occurred as a result of public entities' non-compliance with the provisions of the Public Procurement Act, 2003 (Act 663) as amended (Auditor General report, 2021).

These consequences and losses resulting from sustainable procurement inaction severely ripple the sector's overall sustainable adoption and performance (Ryu & Sueyosh, 2021). The inability of businesses owned by local, women, youth and persons with disability to access contracts and huge financial losses from procurement irregularities is a bane to achieving SDG goals, 1, 2, 5, 8 and 10 (Treviño-Lozano, 2021). It is in the interest of these goals that stakeholders and institutions have been exerting pressures on public procurement entities to adopt and implement sustainable

procurement to increase transparency and cost control through a broader approach to whole-life costing, environmental and social legislation, risk management, and incorporating the local business community in its operations and activities (Agyekum et al., 2022; Adjei-Bamfo et al., 2022).

Although, procurement irregularities in the public sector pose huge financial and social threats to developing economies' wellbeing and economic development (Nyantakyi, 2018; Adjei-Bamfo et al., 2019). Studies on sustainable public procurement are scanty in most developing economies, including Ghana. It is so because previous studies on public procurement have focused on the economic benefits of procurement (Quashie, 2019; Essel, 2021; Asare, 2022). Also, other authors have discussed the literature on sustainable public procurement more qualitatively (Adjei, 2010; Mensah &Ameyaw, 2012; Nyantakyi, 2018; Adjei-Bamfo et al., 2019). Limited studies have investigated the influence of isomorphic pressures, sustainable procurement adoption, and sustainable performance. The study addresses this gap by applying structural equation modeling to examine the influence of isomorphic pressures, sustainable procurement adoption, and sustainable performance of public

1.2 Purpose of the Study

The research examined the influence of isomorphic pressures on sustainable public procurement adoption and performance in Ghana.

1.3 Research Objectives

The specific objectives of the study were to:

i. examine how isomorphic pressures influence sustainable procurement adoption in Ghana.

- ii. examine the effect of sustainable procurement adoption on sustainable performance of public procurement entities in Ghana.
- iii. assess the mediating role of sustainable procurement adoption in the relationship between isomorphic pressures and sustainable performance in Ghana.

1.4 Research Questions

The following questions were formulated in relation to the objectives:

- 1. What is the relationship between isomorphic pressures and sustainable procurement adoption in Ghana?
- 2. What is the effect of sustainable procurement adoption on sustainable performance of public procurement entities in Ghana?
- 3. How does sustainable procurement adoption mediate the link between isomorphic pressures and sustainable performance in Ghana?

1.5 Significance of the Study

Policymakers, as well as public entities interested in the adoption and implementation of sustainable public procurement, will benefit from the study's findings. The results of this study will offer policymakers empirically-based insights, which could result in the creation or improvement of policies that promote sustainable procurement practises. Policymakers can determine the most impactful external forces and subsequently customise their methods to strengthen these pressures. The research findings offer valuable insights into implementing effective strategies for organisations operating in Ghana and similar socio-economic situations. This phenomenon encompasses significant environmental and social ramifications and holds the potential to yield enduring economic advantages for enterprises. Additionally, the study findings can be utilized to supplement empirical and methodological research. Based on the study's findings, there is a link between isomorphism and sustainable public procurement adoption in developing countries, especially Ghana. Also, the utilisation of isomorphic pressures, namely coercive, mimetic, and normative, may be traced back to institutional theory. Examining these forces within the Ghanaian context can yield a deeper understanding of the applicability of global theories within different local environments.

1.6 Delimitations

The research is conducted within the procurement and supply chain setting. The study focused on how isomorphism can influence public procurement entities in Ghana's decision to adopt sustainable procurement practices. The study is based on public procurement entities in Ghana. The study surveyed public procurement entities in Ghana on sustainable public procurement adoption. The entire country was chosen because there are few public procurement entities and the need for a perfect representation of issues. In terms of methodology, the study used a quantitative approach, with data collected using questionnaires. Given each public procurement entity a chance to answer just one questionnaire.

1.7 Limitations

The study adopted the quantitative research approach therefore, questionnaires were used for data collection. This approach eschewed qualitative responses because structured questionnaires restricted the respondents to close-ended questions. The close-ended Likert-type scale statements restricted the quantity and quality of data from the respondents. In addition, there were difficulties in accessing information because some respondents were difficult to locate, as a sample frame of all public procurement entities in Ghana was not appropriately readily available, but the total number given by the public procurement Authority.

1.8 Definition of Terms

- 1. *Isomorphic pressures*: pressures on organisations to conform to their institutional environment due to cohesion from political institution and regulatory bodies, norms of occupational and professional group, and mimicry of legitimized organisations within the industry.
- 2. *Coercive pressures*: pressure from dominant stakeholders that compel public entities to modify their behaviour and practice to those stakeholders applying the pressure.
- 3. *Mimetic pressures*: the pressure to copy legitimized sustainable practices of other public entities in other countries or sectors in Ghana in the face of uncertainties
- 4. *Normative pressures*: pressure from professional bodies, institutions, citizens and professional associations key personnel of public entities are associated with.
- 5. *Sustainable public procurement adoption*: Public procurement actions that result in better environmental outcomes, financial savings, increased consumer surplus, and good social benefits for suppliers and citizens.
- 6. *Sustainable performance*: The ability of public entities to meet the (social, environmental and economic) long-term needs and expectations of suppliers or customers, citizens, and other stakeholders, balanced by an effective management organization through appropriate reforms and innovation in public procurement activities.

7. *Public procurement entity*: procurement entity is defined as any entity conducting public procurement under Act 663 as amended (Section 98).

1.9 Organisation of the Study

There are five chapters that make up the research. Chapter one was all about the study's introduction, which includes the background, statement of the problem, objectives, and hypotheses scope of the study, the study's significance, and how it will be arranged. Chapter Two provides comprehensive reviews of relevant literature to the study. It provides a theoretical and empirical perspective of the study. Chapter Three presents the research method. This chapter provides a methodology that includes the research design, the population, sampling and sampling procedures, the research instrument, and the data collection instrument and data analysis. Also, the sample and data parameter estimates used were described. Chapter Four presents the result and discussion, which discusses the analysis of the findings and interpretation of the data produced or gathered. Chapter Five presents the summary, conclusions, and recommendations. Suggestions for further studies are also provided in the last chapter.

CHAPTER TWO

LITRATURE REVIEW

2.0 Introduction

This chapter contains a review that guides the study concerning the research problem. Views from this chapter are gathered from various writers and researchers on issues related to isomorphism and sustainable public procurement adoption in Ghana. It is organized under different sections; the theoretical review, the conceptual review, and a review of related studies empirically and presents the study's conceptual framework.

2.1.0 Conceptual Review

This session presents a review of the concepts under study. It focuses on the key constructs, including isomorphic pressures (coercive, mimetic and normative pressures), sustainable public procurement adoption and sustainable performance. Out of this review, the conceptual framework for the study was developed.

2.1.1 Isomorphic Pressures

Using the institutional approach, it has been shown that sustainable practices are remarkably similar across different businesses and industries. These practices have been brought to bear by external actors rather than organisational structures (Daddi et.al, 2016). In their paper, Shunbham et.al, (2018) and Raj et al., (2020) explained that these external forces may include; competitors, governments, investment agencies, professional bodies, trade association, society and consumers etc. All these forces place value on sustainability and best practices, which when adopted by firms and entities, promise sustainable performance and legitimacy with stakeholders.

Coercive, mimetic, and normative isomorphic forces influence the acceptance of industrial strategies, including sustainability, IoT, and digitisation. (Powell &

DiMaggio 1983; Li & Ding 2013). The three isomorphic forces of regulatory agency coercion, political influence, legislative rulings, and external player legitimacy. This pressure reflects regulators' demands for eco-friendly manufacturing and consumer behaviour. Influencing how public procurement organizations allocate items and contracts (Acquah et.al., 2021). Due to legislative support and consequences for not complying, many organizations have adopted best practices in public procurement, green manufacturing, and other areas.

Mimetic pressure is generated by organisations normal responses to uncertainty. Companies look to other noble and successful rivals in an uncertain commercial climate to obtain legitimacy; professionals generate normative pressure. In an effort to promote environmentally friendly and responsible behaviour, industry groups and professional bodies frequently implement their green standards and guidelines. This study explores the impact of three isomorphic forces on sustainable public procurement in Ghana.

2.1.2 Coercive Pressures

According to DiMaggio (1983), coercive pressures are present when dominant organisations coerce less powerful ones to follow operational directions. Coercive pressures include government demands, environmental and ethical concerns, cultural norms, dependency on financial resources, and tax law obligations. Organizations under the influence of coercive pressure are those that are dependent on other organisations or societies in the environment in which they operate, according to Somjai and Jermsittiparsert (2019).

Pressure from activists, NGOs, and customers has significantly promoted CSR (Castka & Balzarova, 2008). Coercive pressure can come from governments, companies, and social movements (Kauppi & Hannibal, 2017). Haques (2020) claimed that international corporations may coerce individuals who rely on them to accept

global quality standards. Papadimitriou (2011) argues that organizations are compelled or choose to comply with institutional constraints because of their primary rationality to mobilise resources. This contradicts DiMaggio's (1983) claim that concentrated resources increase the possibility for coercive pressures. Coercive pressure in organizations is key to maintaining financial stability within an institutional framework (Powell & DiMaggio, 1991).

2.1.3 Normative Pressures

DiMaggio and Powell (1983) linked normative pressures to new rules, professionalism, and networking. Employee interactions with other organizations or specific training from a learning institution may impact an organization's structure, operations, and programs (i.e., university or college). In the context of appropriateness, isomorphism is linked to normative isomorphism, which encompasses trade organizations, professional groups, and certification authorities (Heugens & Lander, 2009).

Participation in other groups helps convey professional experience (Kauppi & Hannibal, 2017). According to Kauppi and Hannibal (2017), normative pressures concern organizational domains gaining shared responsibility over field and profession activity. These create acceptable norms of conduct, and stakeholder and institution requests to follow them can drive creative and best practices (Smelt, 2021). Toinpre, Mackee, and Gajendran (2018) postulated that, professional experiences build professional networks that span organizations and propagate new models. This means career chances can transfer professional experiences from one business to another. As long as they hire professionals, similar companies will have similar marketing strategies, policies, procedures, and strategic goals. This aligns organizations.

2.1.4 Mimetic Pressures

When one organization tries to gain legitimacy by copying another organization's behaviour, structure, and decisions, this is called "mimetic pressure" (DiMaggio & Powell, 1983; DiMaggio, 1983). Jaja et al. (2019) found that when an organization is under the same pressure, it builds the same structures, operations, and programs. Here, organizations try to get legitimacy by copying the operations and activities of other organizations. Legitimacy can make it easier for organizations to get more resources (Tsinopoulos et al., 2018; Acquah et al., 2021). This is especially important for non-profits, which means government organizations. Also, mimetic pressures can come about when the operations of focal organizations are unclear (Leiter, 2005).

Uncertainty could come about, for example, if your long-term and short-term goals are unclear. According to Leiter (2005) businesses in the same field tend to copy the operations, structures, and programs of other, more successful businesses in the same field. Galaskiewicz and Wasserman (1989) also found that social networks between businesses in the same area may be a factor in the rise of mimetic pressures. Social networks are made when people from different organizations talk to each other formally or informally. Let's say a manager doubts how or what their organization will do in a certain area. In these situations, the manager might officially ask for help from another organization's management that has solved the same problem.

2.1.5 Sustainable Public Procurement Adoption

Adoption of sustainable public procurement (SPP) has emerged as a key component of sustainability in recent years (Cheng et al., 2018). Green procurement and SPP can potentially address environmental and social concerns through public procurement efforts (Sonnichsen & Clement, 2019). SPP comprises incorporating

environmental, sociocultural, and economic concerns into government or private-sector purchases of commodities, works, and services (Walker & Brammer, 2012; Uttam & Roos, 2015). Sustainable public procurement is the process by which public institutions meet their needs for goods, services, and work to benefit the organization, society, and economy while avoiding environmental damage (Walker & Brammer, 2012; Dza et al., 2021). SPP evaluates design, non-renewable material consumption, manufacturing and production processes, logistics, service delivery, use, operation, maintenance, reuse, recycling choices, and disposal, and suppliers' ability to handle these issues. Core values are transparency, fairness (to the provider and society), long-term economic viability, and responsibility (from purchaser and supplier)

Isomorphism has emerged as a powerful tool in operations management for explaining what motivates organizations to adopt new practices (Dubey et al., 2017; Grob & Benn, 2014), particularly in the context of the environment and Corporate Social Responsibility (Zhu & Sarkis, 2007; Grob & Benn, 2014; Hsu et al., 2014; Seow et al., 2014). Similarly, several supply chain practices, such as electronic supply chain system adoption, green manufacturing adoption, supply chain security adoption, and supply chain performance measurement systems, have all been modelled through institutional isomorphism in recent years (Zhu & Sarkis,2007; Grob & Benn, 2014; Blome et al., 2014; Hsu et al., 2014; Dubey et al., 2017; Liu et al., 2018; Amoako,2021; Acquah et al., 2021).

2.1.6 Sustainable Performance

According to Badi and Murtagh (2019), sustainable performance is detailed research into challenges in both developing and developed countries, including the potential negative consequences of corporate activities on the environment. Meeting fundamental human needs, is a crucial component of defining societal well-being

(Kamble, Gunasekaran, & Gawankar, 2020), as is the preservation of environmental renewable and non-renewable resources for future generations (Gong, Simpson, Koh & Tan, 2018; Poltronieri, Ganga & Gerolamo, 2019). Economic, environmental, and social consequences can all be used to assess sustainability (Ding, Zhang, Li, Tang & Yang, 2018; Zaid, Jaaron & Bon, 2018; San Ong, Magsi & Burgess, 2019).

At an exponential rate, sustainable performance has grown. It is often used in supply chain management because of a more public awareness, strict government regulations, and pressure from the market (Repar, Jan, Nemecek, Dux & Doluschitz, 2018; De Nadae, Carvalho & Vieira, 2019). Many businesses plan their operations to be sustainable (Kusi-Sarpong, Sarkis & Wang, 2016; Bai, Kusi-Sarpong & Sarkis, 2017). Most academics who tried to understand sustainability performance and its meaning used the relationship between the firm and its stakeholders as a lens (Wood et al., 2021). Stakeholders are people or groups vested in a company's actions (Bai et al., 2021).

Sustainability performance shows how much an entity cares about its stakeholders' well-being. Stakeholders track and evaluate how well a company does at being sustainable. Sustainability performance encourages stakeholder collaboration by ensuring that a business's and its stakeholders' interests are the same. This helps the business reach its goals (Yusliza et al.,2020). A business that can keep going for a long time has good pay systems, a lot of qualified workers, and creative management (San et al.,2020). It also makes it less likely that governments will take unwelcome regulatory actions and attract customers who care about social responsibility (Hillman & Keim, 2001; Kamble et al., 2020).

This claim has led to the conclusion that an enterprise's ability to handle stakeholder demands and the significance of its environmental and social stakeholders impacts its sustainability (Brammer & Walker, 2011; Roman, 2017). Sustainability initiatives can be seen as proactive responses to governmental restrictions (Giacomo et al., 2019), responses to interest group activity (Gouda & Saranga, 2020) or methods to differentiate your brand from competitors when marketing to consumers (Aragon-Correa et al., 2008; Lenox & Eesley, 2009; Hull & Rothenberg, 2008; Delmas, Russo, & Montes-Sancho, 2007).

2.2 Empirical Review

In this section, the study's research objectivity is thoroughly assessed of linked literature, contributing to the goal of analysing current studies by comparing and contrasting respective findings.

2.2.1 Isomorphic Pressures (Coercive, Normative and Mimetic pressures) and Sustainable Public Procurement Adoption

2.2.1.2 Coercive pressure on Sustainable Public Procurement Adoption

Coercive pressure improves organizational processes (Lu et al., 2019). The extensive literature review established that coercive pressure is linked to novel manufacturing processes (Ketchen & Hult, 2007), performance assessment methodologies (Dubey et al., 2017), and social sustainability (Ketchen & Hult, 2018). Government regulations often necessitate sustainable purchasing (Dubey et al., 2015). Coercive pressure is recommended to force SPP adoption in public procurement. Multidonor agencies may pressure beneficiaries to follow SPP.

Through an online survey, Zhang, Watson and Zhang (2016) also led a study that focused on the coercive pressures impact on the adoption of sustainable supply chain management methods. The investigation sought whether supply chain

management may lead to positive environmental and economic outcomes. With structured questionnaires, the study collected data from 146 UK manufacturing firms through convenience sampling, the study found that, an integration of government models pressures the adoption of sustainable supply chain management practices.

Other studies have looked at the isomorphic pressures on the construction industry to embrace best practices such as green procurement, sustainable procurement, and building information modelling (BIM) (Cao et al.,2014; Raj et al.,2020; Acquah et al.,2021; Amoako, 2021). A study by Cao et al. (2014) sought to analyse the impact of isomorphic pressures on BIM adoption in construction projects. As part of a quantitative investigation, 92 Chinese construction projects were studied. The partial least square structural equation modelling program was employed for data processing. The authors found a strong favourable influence of coercive pressures on building information modelling.

Furthermore, Acquah et al. (2020) investigated the effectiveness of isomorphic pressures on adopting green manufacturing. The study chose a Quantitative research design and the PLS-SEM approach for data analysis due to the applicability of predictive relevance. The study, one of the first to expose and discuss the isomorphic pressures on the adoption of best practices, green procurement, green product and process innovation, discovered that the combined impact of coercive, mimetic, and normative pressures strongly influences the adoption of green procurement practices. Another source of coercive pressure is citizen pressure based on their attitudes on sustainable public procurement (Raj, 2020). Public procurement produces artifacts with

citizens who are concerned about SPPA. Citizens are considered crucial SPP drivers

taxpayer funds that citizens ultimately consume. Chiarini et al. (2017) identified

18

(Michelsen and de Boer, 2009). Grandia et al. (2015) discovered that understanding customer perceptions of sustainability is important in private-sector procurement.

2.2.1.3 Normative Pressure on Sustainable Public Procurement Adoption

Organizational normative pressure is caused by group expectations (DiMaggio & Powell, 1983). These expectations are transmitted and gradually become organizational norms via a relational network (Berrone et al., 2013). Firms strive to retain procedural legitimacy by adhering to these criteria. According to Raj et al. (2020), several of India's largest public procurement entities, such as Hindustan Aeronautics Limited, publish integrated reports and annual accounts, including sustainability reporting. Many more have followed in their footsteps. Almost all significant companies worldwide, including central and local government procurement agencies, have documented sustainability efforts in their annual reports in recent years, and some are voluntarily adhering to Global Reporting Initiative (GRI) requirements.

Yank and Kang's (2020) examined the effects of normative pressure, quality management practices, and environmental management practices on corporate performance from an integrated framework viewpoint. The study applied the institutional theory. The study relied on the quantitative research approach. Upon critically assessing the structural interactions among the study's construct, the study found that normative pressure has a strong positive relationship with environmental management practices. However, quality management practices were deemed not to have any significant positive relationship with mimetic pressure. Hence, normative pressure was evidenced not to foster the adoption of quality management practices.

Further, in Acquah et al., (2020), the study aimed to investigate the efficacy of isomorphic pressures on the adoption of green manufacturing. The study adopted the survey research design with PLS-SEM technique for data analysis due to the suitability

of predictive relevance. It was found in the findings of the study that, the composite impact of coercive, mimetic and normative pressures robustly influences the adoption of green procurement practices. That is to say isomorphic factors influenced positively and significantly green procurement practices.

Abdul et al. (2019) analysed coercive, mimetic, and normative isomorphisms on environmental performance. GBS moderated the study. Pakistan researched. Utilized institutional theory. Cross-sectional. Eleven industries responded. They surveyed 219 CEOs. The validity of the research collection instrument examined. Two time periods were used to reduce methodological bias. The construct's reliability and validity were confirmed, allowing further research. The study used quantitative research using an explanatory research design and SEM to analyse the data. We tested hypotheses. The study found that coercive, normative, and mimetic institutional pressures positively affect environmentally friendly company tactics (EFBS). Institutional forces positively affect environmental performance, but the link is weak. Just as some studies are in support of normative pressures having a significant positive impact on best practices; sustainable procurement, green procurement, green supply chain, and environmental friendliness (Dubey et al., 2017; Abdul et al., 2019; Yank & Kang, 2020; Acquah et al., 2021), there are a few whose findings proved no significance impact of normative pressures of isomorphic pressures to BIM, green supply chain, Megaproject social responsibility (MSR) behaviour (Lin & Shen 2012; Jain et al., 2020; Xie et al., 2021). In their study, Xie et al., (2020) sought to analyse the impact of isomorphic pressures (coercive, mimetic and normative) on MSR behaviour, with pure altruistic vales and relational behaviour as moderator and mediator variables respectively.

The study employed a sample size of 334 respondents in China, and with the limitation of sample, the study adopted smart PLS- SEM to test various hypothesized relationships. The study results showed an insignificant impact of normative pressure on MSR behaviour in the direct relationship but a significant impact on MSR behaviour through the mediator variable relational behaviour. Indicating a full mediation of relational behaviour in normative pressures and MSR behaviour. Due to different findings and exposition in the literature, the study relies on the institutional theory and the empirical review to hypothesise that;

2.2.1.4 Mimetic Pressure on Sustainable Public Procurement Adoption

Organizations mimic successful competitors to be successful (Zsidisin et al., 2005). In an uncertain environment, this imitation reduces risks and expenses. These countries' governments profit substantially. Mimetic pressure is positively associated with internet-based supply chains (Ke et al., 2009) and green supply chains (Zhu & Sarkis, 2007). Mimetic pressure has a mixed effect on practice adoption. According to certain research, medical pressures don't affect new practice adoption (Gholami et al., 2013). Coercive pressure dominates the early stages of inception, but normative and mimetic pressure take over later.

Zhu and Sarkis (2007) conducted research in China. Chinese manufacturing enterprises' green supply chain strategies and performance were examined for the moderating impacts of institutional pressures. The research was based on institutional theory. The study employed the quantitative approach to the papers' undertaken. The study used convenience sampling. A total of 341sample sizes were used. The study's scales were adopted from empirical studies conducted over the years; hence their reliability and validity were assured. The survey revealed the following after all statistical diagnoses and tests to verify the study's data. According to the report, Chinese manufacturers have been pressured to embrace Green Supply Chain Management methods. According to the study, coercive and normative pressures increase organizations' environmental performance. Mimetic pressure affects enterprises that use GSCM but not environmental performance. The study hypothesized that no institutional pressure isomorphisms generate "win-win" conditions for manufacturing enterprises.

Furthermore, Saeed et al. (2018) studied whether external and internal green supply chain management strategies face the same or distinct coercive, normative mimetic pressures. The institutional and resource dependence theories were used to guide the investigation. The study was conducted in Pakistan. The study employed a quantitative research technique and a standardized questionnaire to collect data. The data collection instruments for the study were all assessed on a 5-point Likert scale adapted from Zhu et al. (2013). The poll elicited 207 responses from management-level executives in Pakistani manufacturing firms at random. The paper employed Partial Least Squares-Structural Equation Modelling to analyse the study's data (PLS-SEM).

PLS-SEM analysis of the structural and measurement models produced some encouraging results once the study's credibility was established by cross-checking of the processed data's reliability and validity. Green Supply Chain Management techniques were shown to have a considerable influence on the study's internal and external normative pressures. Coercive and mimetic pressures significantly influence both internal and external GSCM practice, contrary to what was previously stated for normative pressures. Internal GSCM activities have less impact on environmental performance than external GSCM practice, similar to normative pressure. Finally, it was determined that manufacturing businesses' external GSCM procedures and internal GSCM practice work in concert. The study relies on the institutional theory and the empirical review to hypothesise that;

2.2.1.5 Sustainable Public Procurement Adoption and Sustainable Performance

Sustainability implementation leads to environmental advantages, cost reduction, consumer surplus, social benefits, etc. (Brammer & Walker, 2011; Roman, 2017). SPPA activities are believed to improve economic, environmental, and social sustainability (Giacomo et al., 2019). Sustainable purchasing may seem more expensive initially, but maintenance and disposal costs are lower in the long run. Sustainability efforts improve performance, according to Gouda and Saranga (2020).

Zhu and Sarkis (2007) argue that adopting green purchasing has helped improve organisational performance in China. Similarly, Mani et al. (2018) show that adopting social sustainability enhances buyer's social reputation and supplier's social performance. Adebayo et al., (2020), using a case study of a company, looked for ways to embed sustainable practices within the company's context. The study's findings revealed that their organizational culture highly influences the environmental performance of Fast-Moving Consumer Goods (FMCGs) companies. Further, found that institutional pressures and significantly influenced internal sustainability capabilities. However, task performance as a subconstruct had the least significant level of environmental performance.

There is no proof that SPPA improves sustainable performance. If adoption increases, procurement groups would prefer suppliers engaged in sustainability initiatives and penalize suppliers for adopting methods that negatively affect the environment and social issues (Cheng et al., 2018; Johnsen et al., 2017). Because of this, customers are inclined to share information about their environmental sustainability efforts with their suppliers and would be prepared to share their suppliers'

sustainability adoption expenses. Witjes and Lozano (2016) suggest a collaborative model between supplier and buyer to boost SP. It may lead to higher sustainable performance. Improved sustainable performance will likely boost economic, environmental, and social performance. The study relies on the institutional theory and the empirical review to hypothesise that;

2.2.1.6 The role of Sustainable Procurement Adoption in the Relationship between Isomorphic Pressures and Sustainable performance

Pressures from stakeholders and institutions with a given regulation may vary considerably across organisations (Debroux, 2010; Gray & Silbey, 2014), and compliance reasons may also vary significantly. Some organisations may comply to avoid punitive sanctions, achieve legitimacy, or protect their license to operate (Barnett & Carroll, 1995; DiMaggio & Powell, 1983; Gunningham, 2010); others may comply because they believe compliance is the right thing to do (Gunningham, 2010). Conversely, some organisations may not comply if they perceive regulation as a threat to the realisation of organisational goals or corporate profitability (Gray & Silbey, 2014; Russell et al., 2008). Lack of awareness or knowledge of a given regulation may also affect compliance (Gray & Silbey, 2014; Zhu et al., 2018). Moreover, compliance with a given regulation may be negatively affected by an organisation's tendency and ability to exploit inherent legal loopholes in a given regulation (Parker et al., 2009).

K. çükoğlu and Pinar (2016) carried out a study to examine the relational effect of sustainability drivers on green innovation. The investigation subsequently assessed the relationship by mediating it with green organisational culture (sustainability culture). The study was undertaken within the Turkish economy and relied on 162 out of the top 500 companies published by the Istanbul Chamber of Industry from 2010 to 2012. It used a quantitative research approach and an explanatory research design.
Researchers found a strong correlation between green innovation and sustainability drivers. Sustainability drivers were shown to have a considerable influence on green innovation. On the other hand, green organizational leadership significantly influences innovation. Green corporate leadership also significantly impacted the statistical link between sustainability drivers and green innovation.

Etse et.al. (2022) noted that an organization's response to a regulation may depend on its nature. Este et al. (2011) studied the impact of rules on sustainable procurement using organizational culture and top management support as mediators. The PLS-SEM analysis of 322 Ghanaian organizations revealed a moderate sustainable procurement practice, a positive effect of regulation on related practices, a mediation effect of organisational top management support on regulation's relationship with sustainable procurement, and no mediation effect relative to organizational culture. Sönnichsen et al. (2020) reviewed green sustainable public procurement literature spanning 2000-2020. Data classifications revealed three themes: organizational, individual, and behavioural. Circular public procurement relates to strategies, requirements, costs, size, and knowledge. Influence and knowledge in organizational leadership are functional prerequisites. (Khan & Aldenius, 2017). Without functional criteria, local governments can't support innovation in specialized technologies. Only the largest area can affect market development, which is significant because market expansion and innovation are key to the European Commission's circular economic strategy.

2.3 Theoretical Review and Hypothesis Development

The study was conducted through the lens of institutional and stakeholder theory due to their relevance to its objectives. This section, therefore, discusses the institutional and stakeholder theories and how they extensively explain the study's objectives.

2.3.1 Institutional Theory

Institutional theory gives a multifaceted account of how organizations become homogeneity under social influences, sometimes from outside sources of the organisation and others from inside (Meyer & Rowan, 1977). These forces might shift an organization's focus away from just economic success and toward widely practiced features, such as environmental consciousness, professional certification and widespread activities in other businesses. The environmental behaviours of an organization have been studied using institutional theory (Mitchell, 2015; Bruton, Ahlstrom, & Li, 2010). It has been noted that institutional theoretical statements have one thing in common: anything discovered at a higher level is utilized to explain processes and outcomes at a lower level of study (Clemens & Cook 1999; Amenta, 2005).

Researchers in operations management have utilized institutional theory to explain why firms adopt new practices (Grob & Benn, 2014; Dubey et al., 2017), especially in the context of the environment and CSR (Zhu & Sarkis, 2007; Grob & Benn, 2014; Hsu et al., 2014; Seow et al., 2014). Researchers have used Institutional theory to examine the adoption of various supply chain practices, including electronic supply chain management system adoption (Ke et al., 2009), supply chain security practices (Liu et al., 2018), green supply chain practices (Blome et al., 2014; Zhu & Sarkis, 2007), and performance measurement systems (Dubey et al., 2017). According to this view, corporations are inevitably bent on following social norms and values (DiMaggio & Powell, 1983; Liu et al., 2010) Unlike other public procurement theories investigated, such as economic quantitative theory (Lu et al., 2019), the institutional theory focuses on organizations' social efforts and how these efforts enhance an organization's reputation, performance, and ability to acquire social resources (Raj et al., 2020). As a result, many companies base their decisions on the social rationale, understanding that sustainability measures in public procurement may restrict short-term efficiency or incur additional expenses. Still, they are crucial in gaining stakeholders' respect in the long run. As a result, the use of Institutional Theory in the context of sustainable public procurement is permissible.

According to the Institutional theory, pressures come from three sources: coercive, normative, and mimetic. Organizations obtain legitimacy, resources, and survival capacities by building structures or adopting isomorphic activities with institutional forces (DiMaggio et al., 1983; Meyer et al., 1977). The growth of institutional isomorphism occurs through three mechanisms: normative, mimetic, and coercive. Experimentally, the three processes are not always distinguishable (DiMaggio et al., 1983; Mizruchi et al., 1999). Compliance under normative pressures occurs when businesses feel required to respect specific cultural norms imposed by professional circles or society. Multilateral and dyadic networks can disseminate normative pressures. Mimetic isomorphism occurs when companies emulate other organizations for legitimacy or established processes. Coercive pressure refers to the pressures an organisation feels to comply with certain rules, norms, or expectations due to formal or informal demands imposed by other organisations or institutions on which it depends or by societal norms. (DiMaggio et al., 1983; Tolbert et al., 1983).

The institutional theory emphasizes the crucial importance of the institutional setting in compelling organizations to undertake changes in structure and behaviour to

obtain social legitimacy (DiMaggio & Powell 1983; Scott, 2001). Institutional theory has compellingly explained various organizational transformations and innovation diffusion in different industries by understanding organizations as open systems vulnerable to the effects of specific surroundings (Teo et al., 2003; Bhakoo & Choi, 2013). In the same way, because sustainable procurement adoption is a global goal linked to the SDGs. Organisations must be made to comply using different approaches. Implementing sustainable procurement practices can yield favourable outcomes by bolstering reputation, cultivating stakeholder relationships, and optimising operational efficiency. Hence, the institutional theory provides the right philosophical background for examining this compliance. Accordingly, in this study, the institutional approach might also give substantial insights into how sustainable public procurement is adopted and implemented in Ghana.

2.3.3 Stakeholder Theory

Every corporate entity lives in an ecosystem of related groups that it must satisfy to be successful (Friedman & Miles, 2006). According to Freeman (1984), a stakeholder is a person, group, or body affected by an entity's short- and long-term actions. According to this viewpoint, a company's success is determined not just by achieving the expectations of its shareholders but also by the people and organizations impacted by its operations. (Freeman, 1984; Gray, 1995). According to the stakeholder theory, a company's operations should be targeted to all its components (Hahn & Kuhnen, 2013; Freudenreich et al., 2020).

The theory recognizes the dynamic and complex nature of an organisation's relationship with its stakeholders, which requires the organization to participate in activities commensurate with stakeholder objectives (Gray et al., 1996). Freeman (1984) stressed that, when the business environment changes or new practices arise,

University of Education, Winneba http://ir.uew.edu.gh

entities must consider the dynamic nature of stakeholders. Adopting sustainable procurement methods shows an entity's commitment to the SDGs (Brammer & Walker, 2011; Adjei-Bamfo et al., 2019; Grandia & Kruyen, 2020). The stakeholder theory drawing inference from the institutional theory and connotes that, adopting best practice with the desire to gain legitimacy originated from diverse stakeholder groups. Indicating that organisations and entities are inseparable from their stakeholders and society.

Stakeholder Theory emphasizes stakeholders' impact on the decision-making and performance of organisations. Within the context of public procurement, various stakeholders, including but not limited to customers, employees, investors, regulators, governments and the general soties, may express a desire or expectation of organizations to prioritize sustainability in their operations and consumptions. Organisations may opt to implement sustainable procurement practises in response to stakeholder demands. This approach can yield benefits such as fostering favourable stakeholder relationships, bolstering brand reputation, and mitigating potential risks to reputation. Adopting sustainable procurement practises by organisations can lead to improved performance, encompassing both financial performance and long-term sustainability, by meeting stakeholders' expectations.

2.4 Hypothesis Development

2.4.1 Isomorphic Pressures and Sustainable Public Procurement Adoption

The institutional theory suggests that organization's structures, routines, and behaviors are heavily influenced by the institutional environment in which they operate. This environment is characterized by the regulatory, normative, and cognitive structures that provide stability and meaning to social behaviors (DiMaggio & Powell, 1983). The Coercive pressure occurs when organizations adopt specific structures or behaviors due to formal pressures, often from government or influential stakeholders (DiMaggio & Powell, 1983). With the increasing global concern for environmental sustainability, there's a surge in regulations and mandates related to sustainable business practices (Bansal & Roth, 2000). Governments and regulatory bodies are pushing organizations to adopt sustainable practices, including procurement (King & Lenox, 2000). Empirical research has shown that coercive pressure enhances adoptions (Zhang, Watson & Zhang, 2016). A study by Raj et al. (2020), Acquah et al. (2021) and Amoako (2021) proved that isomorphic pressure affects the attitudes of firms to adopt sustainability.

Similarly, normative pressure emerges from professionalization and the collective struggles of members of an occupation to define the conditions and methods of their work (DiMaggio & Powell, 1983). As the business world becomes more conscious about sustainability, professional networks, industry associations, and educational institutions emphasize sustainability's importance in procurement (Scott, 2001). As professionals in procurement roles are trained and socialized in this norm, they bring these values to their organizations, promoting sustainable procurement practices. Research by Yank and Kang (2020) revealed that normative pressure has a strong positive relationship with environmental management practices. Abdul et al. (2019) and Acquah et al. (2020) affirmed that normative pressure influences green procurement adoption practices.

Lastly, the mimetic pressure explains how organizations often model themselves after other successful organizations (DiMaggio & Powell, 1983). Suppose industry leaders or major competitors adopt sustainable procurement practices and gain competitive or reputational advantages. In that case, other organizations will likely imitate or emulate those practices to reduce uncertainty and gain similar benefits. Adopting sustainable procurement becomes a benchmark for best practices, leading others to follow suit (Jennings & Zandbergen, 1995). Firms often adopt environmental management systems in response to competitors or industry leaders (Delmas & Toffel, 2008). Research on adopting environmental management systems has shown that firms often adopt these systems in response to the practices of their competitors or industry leaders. Extant literature reveals that mimetic pressure help firms to adopt green supply chain management practices. Abdul et al. (2019) and Acquah et al. (2020) affirmed that mimetic pressure influences green procurement adoption practices. Anchored on these observed empirical and theoretical (institutional theory) based findings, the study proposed that

H1a: Coercive pressures have a significant positive influence on sustainable procurement adoption in Ghana.

H1b: There is a significant positive influence of normative pressures on sustainable procurement adoption in Ghana.

H1c: Mimetic pressures have a significant positive influence on sustainable procurement adoption in Ghana.

2.4.2 Isomorphic Pressures, Sustainable Procurement Adoption and Sustainable performance

The stakeholder theory, as proposed by Freeman (1984), posits that the success of an organization depends on the extent to which it meets the needs and addresses the concerns of its stakeholders. In today's business environment, stakeholders increasingly expect organizations to prioritize sustainability. Sustainable procurement practices, which ensure that goods and services are sourced responsibly, address these expectations directly (Freeman et al., 2004). Organizations are better positioned to achieve sustainable performance outcomes by aligning procurement practices with stakeholder values. Also, sustainable procurement offers a competitive advantage by reducing risks, ensuring long-term availability of resources, and meeting customer demands for ethically sourced products (Porter & Kramer, 2006). As organizations integrate sustainability into their procurement processes, they are more likely to enhance their reputation and gain stakeholder trust, leading to improved sustainable performance. Studies have shown that engaging with stakeholders and addressing their sustainability concerns positively affects organizational performance (Hillman & Keim, 2001; Wnager, 2011). Based on precedence, the study relies on the stakeholder theory and the empirical review to hypothesise that;

H3: Sustainable procurement adoption plays a significant role in Ghana's relationship between isomorphic pressure and sustainable performance.

2.5 Conceptual Framework

A conceptual framework is a collection of concepts assembled as a map for a study to demonstrate the link between the researcher's variables of interest (Mugenda & Mugenda, 2008). The framework was created expressly to explain the link between the study's independent and dependent variables. Figure 1 depicts the conceptual framework of the study.



Figure 1: Conceptual Framework Source: Author's Construct, (2023)

The independent variable (Isomorphic pressures) predicts the degree of variance in the dependent variable (Sustainable performance), as shown (Konthari, 2008). The value or degree of sustainable public procurement adoption is affected by changes in the isomorphic forces, which include coercive, normative, and mimetic pressures. Finally, an endogenous variable of sustainable performance was measured using all three aspects of sustainability: economic, environmental, and social. The framework was then created to visually represent the link between the various isomorphic forces and the adoption of sustainable public procurement in Ghana, from the diagram figure 1. The dotted lines depict a mediating relationship between isomorphic pressures and sustainable performance through sustainable procurement adoption.

2.6 Chapter Summary

This chapter carefully analysed relevant literature to support the study's theories: institutional theory and stakeholder theory. Additionally, the research conceptual framework was presented, as was a discussion of several concepts under the

University of Education, Winneba http://ir.uew.edu.gh

conceptual review section and empirical findings of other researchers. The next chapter explains the research methods employed in the study.



CHAPTER THREE

RESEARCH METHODS

3.0 Introduction

This chapter discusses the research approach used to explore the research problem under consideration. According to Kothari (2004), a research methodology is a practical approach to discovering solutions to research questions. This chapter focuses on the methods employed in this investigation. The research design, population and sampling processes, sample size, data collection instruments, data collection procedures, and data processing and analysis were all discussed in this part.

3.1 Research Philosophy

Kuhn first used the term research paradigm in 1962. According to Kuhn, a research paradigm is a collection of scientifically agreed-upon principles for analysing and addressing problems (Kuhn, 1970). The research paradigm guides the decision of researchers. It helps researchers to define a research problem carefully, formulate objectives and research questions, and determine the research's reality, methodological approach, and knowledge base (Kivunja & Kuyini, 2017; Khatri, 2020). The paradigm consists of four components: ontology (what we think about the world), axiology (ethical implications to be considered when conducting research), methodology (a valid and very well approach to data collection), and epistemology (knowledge and how it is obtained and tested) (Kivunja & Kivunja, 2017; Aliyu *et al.*, 2015; Mertens, 2010).

Denzin & Licoln (2018) grouped the paradigms into seven categories: feminist, positivism/postpositivism, Marxist, ethnic, constructivism, queer theory, and cultural studies. The study's goal falls within the positivist approach because a hypothesis can be developed and tested objectively. The specific objective of this study requires gathering data to generalise the results to the entire population. The positivist paradigm enables social scientists to present their research as experts in a detailed, systematic, and systematic way (Hasan, 2016; Benton & Craib, 2010). According to positivists, research is conducted independently of society, and phenomena are evaluated using objective standards. It originates from epistemological conventions that a "belief is an external reality constituted of facts structured in law–like manner" (Evered & Louis, 1991).

The ultimate focus of the positivist inquiry is to reveal causal relations or explanatory relationships that result in a prediction of a phenomenon being studied (Sciarra, 1999). The positivism paradigm holds that research is suitable if it has both internal and external validity and is objective and reliable (Rehman & Alharthi, 2016). In order to gather information, positivists typically employ quantitative research techniques. Under the positivist paradigm, the study's results can be extrapolated (Alharahsheh & Pius, 2020). The positivist research paradigm was used to assist the researcher in gathering data, analysing it, and presenting the results unbiasedly to establish a relationship between the variables The study's findings can be generalised to the entire population under the positivist paradigm (Alharahsheh & Pius, 2020).

3.2 Research Approach

Research approach refers to the overall strategy against which specific designs are implemented. This study employed the quantitative research approach in line with the positivism paradigm. Quantitative research involves collecting and analysing numerical data and applying statistical text. It is a means for testing objectives and theories by examining the relationship among variables. Again, the quantitative approach is suitable for generalisation (Bresz & Creswell, 2017). If one wants to know the impact of an intervention or a policy on a larger population. Quantitative data is scientific, fast and draws on logical conclusions from numerical values obtained from surveys and questionnaires. It is relevant for examining cause and effect relationships among variables (Creswell & Creswell, 2017). However, the quantitative approach has been criticised for its inability to gauge human behaviour (Crotty,1998) effectively. Despite these weaknesses, the quantitative approach was adopted due to the research philosophy and nature of the research objectives, establishing cause and effect relationships among the variables among variables comprising isomorphic pressures such as coercive, normative, mimetic and sustainable procurement adoption.

3.3 Research Design

The research design could be exploratory, descriptive, or explanatory. Williams (2011) suggests choosing a study design based on the research environment, challenge, potential restrictions, and guiding paradigm. The study employed an explanatory research approach to analyse knowledge, beliefs, preferences, and concepts on the relationship between isomorphic pressures and sustainable public procurement adoption. Explanatory research improve comprehension of a subject and help generalize findings (Creswell, 2014). Explanatory studies provide social workers with various knowledge about subjects, contexts, and surroundings (Siedlecki, 2020).

Explanatory design explains the relationship between phenomena. Using descriptive and inferential statistical tools, the approach gathers and analyses massive volumes of data from a diverse target population (Tabachnick & Fidell, 2007). The explanatory design employed a structured questionnaire to collect data from respondents. The concept is suitable for gathering data from public procurement bodies nationwide. Again, the explanation uses statistical tools to analyse cause-and-effect links between variables (McCarthy, 2017). The design was used to establish a cause-

and-effect link between variables. Explanatory design is best for testing isomorphic pressures and sustainable public procurement adoption.

3.4 Study Area

The study was conducted focusing on public procurement entities in Ghana. The public procurement entities were selected because government entities are considered both centers of production and consumption (source). As a result, great contributors to greenhouse gas (GHG) emissions and non-renewable energy consumption. And there is a need to adopt and mainstream sustainability in the procurement process. The novelty nature of public procurement in Ghana and the aim of collecting data from procurement officers of each procurement entity warranted the study to focus on all public procurement entities in Ghana. The study focused on the procurement practitioners of public procurement entities in Ghana as its unit of observation and public procurement entities as its unit of analysis. Hence, every questionnaire was administered to a public procurement officer of every single procurement entity visited by the researcher.

3.5 Study Population

According to Leedy and Ormrod (2010), the population is the target group for which the researcher is seeking information and generating conclusions. A population is a group of people, events, or objects researchers study by observing certain qualities (Saunders et al., 2009). The study's target population was all public procurement entities in Ghana. Thus, the analysis unit was procurement practitioners of public procurement entities in Ghana. These key personnel were chosen due to their philosophies, values and direct involvement in procuring goods, works and services in their respective entities. Thus, their abilities to influence procurement decisions would help obtain relevant information to draw conclusions to the study's objectives. According to the public procurement authority of Ghana PPA (2021), there are 660 public procurement entities in Ghana. This includes all Ministries, Departments and Agencies (MDAs), Metropolitan, Municipal and District Assemblies (MMDAs), State Owned Enterprises, Hospitals and Tertiary institutions.

3.6 Sample and Sampling Procedure

A sample represents the number of respondents in the population that will represent the entire group (Mugenda, 2003; Saunders et al., 2009). Samples are drawn to represent the entire population. (Saunders et al., 2009). The sample size is a subset of the population drawn to represent the entire population (Garson, 2012). The study used a simple random sampling technique. Simple random sampling ensures that every population element has an equal chance of selection. According to Fraenkel et al. (2012), simple random sampling is the most effective technique for extracting a representative sample applied to large samples.

The simple random was deemed appropriate because of the homogeneity of the entities and the availability of the sample frame (thus, the public entities). Three stages went into the selection of the sample. The study's first step identified and compiled a list of public entities in Ghana. In the second stage, numbers were assigned to the public entities. Finally, participants in the study were chosen randomly from a pool of applicants. These stages were followed to ensure that each candidate had an equal chance of being selected.

Smith (2010), suggests that the sample size is the amount of observation used to predict the entire population. The sample size is a larger population subgroup (Saunders, Lewis & Thornhill, 2012). The Krejcie and Morgan (1970) sample size estimation table suggested 249 respondents out of a population of 660 public procurement entities in Ghana as a minimum sample size. To ensure accuracy and reliability, the study gathered data from one key representative of 330 public procurement entities out of 660 entities. The rationale for selecting key personnel as respondents was because their values, job positions and ideologies directly influence procurements, providing relatively accurate information to the study.

3.7 Data Collection Instruments

A primary data collection instrument, more precisely a structured questionnaire, was employed in this study. This was appropriate because of the research approach and research design used. Explanatory study designs are naturally structured. Therefore organised primary data collection procedures are required (Maxwell, 2012). A questionnaire is a tool for collecting data that consists of closed-ended and open-ended questions given to research respondents to acquire the respondents' objective thoughts on a subject (Singer & Couper, 2017). Similarly, Malhorta and Briks (2007) defined a questionnaire as a collection of questions approved for use by respondents in the process of data collection.

Questionnaires are the most common data-gathering tool in research and can measure issues crucial to business and growth (Malhotra & Birks, 2007). A questionnaire is suitable for gathering information from a large number of people. It aids in collecting objective data from a large group to ensure reliable and valid responses. The individual items on the questionnaire were carefully extracted from validated literature. The questions were adapted from extant literature as indicated in Table 1. The questions took 30 minutes to complete. All responders answered the same questions anonymously. The questionnaire was structured into five sections. The structure was based on the objectives of this study. Section "A" asked about coercive, normative, and mimetic pressures from stakeholders, governments, and consumers to embrace sustainable public procurement. Section "B" asked about sustainable public procurement in Ghana. Section "C" asks about sustainable performance, considering environmental, social, and economic factors. Section "D" covered the respondents' demographics.

3.8 Ethical Consideration

Ethics are the norms or standards that impact moral behavior and relationships (Cooper & Schindler, 2008). Study ethics involves developing a moral and ethical research topic and communicating the findings. According to Malhotra and Birks (2007), research ethics involve respondents and the public. The researcher carefully followed the study and ethical guidelines to avoid causing any form of harm to the respondents. Public procurement entities received an introductory letter from the researcher from the Department of Procurement and Supply Chain Management. The researcher duly followed all ethical procedures in researching the influence of isomorphic pressures on sustainable public procurement adoption in Ghana.

In addition, the respondents were notified that their responses were voluntary and that they might opt out at any time. However, they were encouraged to complete the survey in its entirety. Consequently, confidentiality, subject autonomy, and anonymity were upheld throughout the study. In addition, the questionnaire was designed with respondent confidentiality in mind.

3.9 Operationalisation of Key Variables

This section described how the study variables were measured to fulfil research aims. All of the measuring items were based on sustainable public procurement literature reviews. Table 1 revealed all variable indicators were based on in-depth literature reviews. This is because the table shows the different places from which the indicators were taken. But it is still not clear how well these things are measured. As a result, a pre-test was done on a small number of Ghanaian public procurement entities. This was done to see if the indicators of the different constructs were good quality measures in the context of isomorphic pressures and sustainable public procurement. The measures are shown in table 1 below.

Variables	Measurement Items	Sources
Coercive	Coercive pressure was defined as pressure	(Liu et al., 2010;
pressures	exerted by dominant stakeholders on public	Roxas & Coetzer,
	entities to adjust their behaviour and	2012; Lu et al., 2019)
	practices to those acceptable to those	
	applying the pressure.	
Normative	Normative pressure was operationalized and	(DiMaggio & Powell,
pressures	defined as pressure from professional	2002; Berrone et al.,
	bodies, institutions, citizens and	2013; Latan et
	professional associations key personnel of	al.,2018)
	public entities are associated with (eg.	
	CIPS, ISM, CILT, ASCM)	
Mimetic	Mimetic isomorphism was operationally	(DiMaggio & Powell,
pressures	defined as the pressure to copy legitimized	2002; Zsidisin et al.,
	sustainable practices of other public entities	2005; Latan et
	in other countries or sectors in the face of	al.,2018;)
	uncertainties	
Sustainable	Public procurement activities that lead to	(Brammer & Walker,
public	favourable environmental outcomes, cost	2011; Walker, 2015;
procurement	reduction, better consumer surplus and	Witjes & Lozano,
Adoption	positive social benefits of suppliers and	2016; Leal-Filho et
	citizens	al., 2019)
Sustainable	The ability of public entities to meet the	(Carter & Jennings,
performance	(social, environmental and economic) needs	2004)
	and expectations of suppliers or customers,	
	citizens and other stakeholders.	

Table 1:Measurement of Variables and Sources

Sources: Field survey (2023)

3.10 Data collection Procedure

The Department of Procurement and Supply Chain Management provided an authority note, which was appended to the questionnaire distributed to all public procurement institutions. The data was collected via a self-administered questionnaire devised by the researcher to guarantee a high response rate. The data was standardized so that each respondent received identical questions. Respondents were encouraged to answer the questions as thoroughly as possible. This provided respondents a certain amount of time to respond to the questions, and after a week, the researcher and field assistants returned to respondents and collected the completed questionnaire.

The data collection exercise had some associated challenges, including respondents' unwillingness to answer the questions. These challenges were, however minimised by providing introductory letters acquired from the department giving assurance of ethical clearance and the need for the activity for academic purposes only. Again, respondents who wanted to complete the questionnaire at their convenience were allowed to do so.

3.11 Data Processing and Analysis

Data processing and analysis reduce enormous amounts of data into meaningful information and reports for decision-making and policy formulation. Cooper and Schindler (2008) define data analysis as reducing and modifying acquired data, summarizing, seeking patterns, and utilizing statistical approaches. These approaches improve data analysis accuracy while guaranteeing assumptions are not broken (Tabachnick & Fidell, 2001; Hair et al.,2011). SPSS version 26 was used to code and process data. The processed data were analysed using descriptive and inferential techniques. The descriptive comprise, percentages, means, frequencies, standard deviations, kurtosis and skewness. Whereas the inferential technique employed was the Partial Least Squares- Structural Equation Modelling (PLS-SEM 4.0.9) to achieve all the research objectives.

PLS-SEM combines factor analysis and multiple regression by minimising the residual variances of the endogenous constructs (Hair et al., 2011). The technique

develops more precise estimates of factor scores because the PLS algorithm calculates latent variable scores as exact linear combinations of the observed indicator variables, which serve as proxies for latent variables, in order to estimate model relationship (Lowry & Gaskin, 2014). On this note, the hypothesized relationships, most notably between the latent variables, can only be meaningfully interpreted if construct validity was established (Peter & Churchill, 1986). Researchers ensure that the measurement models capture what they intend to measure (Campbell & Fiske, 1959). Specifically, the PLS-SEM algorithm first optimizes the measurement model parameters and then, in a second step, estimates the path coefficients in the structural model. Thus, researchers applying PLS-SEM must first examine the measurement models' characteristics by assessing the reliability and validity of the indicators and eliminating those that may fall out of the acceptable threshold. According to Hair et al. (2018), the key areas examined for reliability and validity include indicator reliability, internal consistency reliability, convergent and discriminant validity.

After meeting the fundamental assumptions of multicollinearity, indicator and construct reliability, discriminant and convergent validity, and outer model significance, the study hypotheses were assessed using PLS-SEM. PLS-SEM was used because it allows researchers to estimate complex models with multiple constructs, indicator variables, and structural paths without imposing distribution assumptions on the data, enabling the creation of inferential statistics such as hypothesis testing. Purwanto and Sudargini (2021) believe examining indicator loadings are the first step in evaluating the reflective measurement model. Authors propose item loadings greater than 0.708 because they show that the structure explains more than half of the indicator's fluctuation and provides appropriate reliability. The second phase of PLS-

SEM is to analyse internal consistency dependability, most typically using Joreskog's

(1971) composite reliability, with a larger value indicating a better level of reliability. Purwanto and Sudargini (2021) consider a 0.60 to 0.70 reliability rating acceptable, and 0.70 to 0.90 "excellent". A 0.95 dependability grade suggests an undesirable or unwelcome reaction pattern. The third phase in evaluating the reflective measurement model, according to Hair et al., (2019), is a consideration of constructs convergent validity. Convergent validity was defined by Purwanto and Sudargini (2021) as the degree to which the constructs converge to explain the variance of the items. The (AVE) was also mentioned by the authors as the metric used by researchers to examine the convergent validity for items in each concept. In addition, Ramrez and Palos-Sánchez (2018) regarded the AVE as the most commonly used metric of analysing convergent validity in PLS-SEM. Hair et al. (2019) discovered that an AVE value of 0.50 or higher is acceptable, indicating that the construct explains or accounts for at least 50% of the variation of the items.

Fourth, evaluate the reflective measurement model's discriminant validity, suggest by Purwanto and Sudargini (2021) and Hair et al. (2019). Hair et al. (2019) argue discriminant validity reveals how different a model construct is from others. Henseler et al. (2015) found Fornell and Larcker's (1981) criterion to be unreliable over time. They used HTMT to measure discriminant validity. HTMT compares two latent variables (Henseler, 2017). The Henseler et al. (2015) HTMT cut-off is 90. The authors indicated discriminant validity exists if HTMT is high, >90, especially if the conceptions are conceptually dissimilar. To ensure unbiased regression results, collinearity is checked before assessing the structure of the relationships, and to examine the collinearity, the variance inflation factors (VIF) is calculated (Purwanto & Sudargini, 2021). VIFs measure the induced collinearity in the effects (Craney & Surles, 2002). According to Purwanto and Sudargini (2021), an ideal VIF value should

be less than or equal to $3(VIF \le 3)$. VIF values above five (5) indicate that there is a possible collinearity problem among the predicting constructs (Purwanto & Sudargini (2021). Hair et al. (2019) specified that if the model is free from collinearity, the next step is to assess the endogenous construct's R Square (R2).

Rigdon (2012) said the R^2 is the sample's prediction power. Purwanto and Sudargini (2021) found that the R^2 ranges from 0 to 1; higher values are thought to have more power to explain in the sample. Henseler et al. (2019) say that values of 0.75, 0.50, and 0.25 are strong, moderate, and weak, respectively. Hair et al. (2019) wrote that researchers can figure out how removing some predictor constructs affects the endogenous constructs' f Square (f2) value. Cohen's (1988) rule of thumb says that values greater than 0.02, 0.15, and 0.35 mean the effect size is small, medium, or large.

3.12 Chapter Summary

The study examined the influence of isomorphic pressure on sustainable performance in Ghana and the role of sustainable procurement adoption as a mediator. Respondents were recruited via simple random sampling technique for an explanatory survey. Self-administered surveys were used to in gathering data. SMART PLS (Version 4) was used for data analysis. This chapter examined the experiment's measuring instrument's reliability and the researcher's ethics. It stipulates that respondents' identities were protected and results were only utilized academically.

CHAPTER FOUR

RESULTS AND DISCUSSION

4.0 Introduction

This study examines the impact of isomorphic pressures on sustainable public procurement in Ghana. This chapter describes respondent demographics, data presentation, and the three research objectives analysed using structural equation modelling (SEM) via the partial least square (PLS) approach.

4.1 Response Rate

The Krejcie and Morgan (1970) sample size estimation table suggested a minimum of 249 public procurement entities in Ghana. To ensure accuracy and reliability, the researcher distributed 330 questionnaires to the entities. Out of the 330 questionnires distributed, 311 were retrieved, but 4 (1.2%) were incomplete, hence were not included in the analysis. The study analysis relied on the 307 usable questionnaires, representing 93.0%.

4.2 Demographic Characteristics of Respondents

The demographic characteristics of the respondents were presented using frequency and percentage. The analysis looked at respondents' demographics, where they were asked about their gender, position in the organization where they worked, number of years they had been there, highest degree of education, professional bodies to which they belonged, and sector in which they worked.

Characteristic	Category	Frequency (N)	Percentages (%)
Sex	Male	203	66.1
	Female	104	33.9
Position	Procurement officer	141	45.9
	Assist.Procurement	95	31.0
	Store Keeper	71	23.1
Years of Experience	1-10 years	129	42.0
	11-20 years	145	47.2
	21-30 years	32	10.4
	Above 30 Years	1	.3
Educational Qualification	Diploma/HND	93	30.3
	Bachelor	135	44.0
	Masters	74	24.1
	PhD Other	5	1.6
Professional Affiliation	CIPS O CIPS	166	54.1
	CILT	82	26.7
	GIPS	16	5.2
	ACCA	1	.3
	ICAG	2	.7
	Other	40	13.0
Sector	MDAs	16	5.2
	MMDAs	119	38.8
	State Owned	69	22.5
	Enterprises		
	Hospital	82	26.7
	Tertiary Institutions	21	6.8
	Total	307	100

Source: Field survey (2023)

The demographic analysis revealed that men (203) made up the majority of the research group (66.1%), while females (104) made up 33.9%. This reflects the nature of male domination in public procurement entities in Ghana and reinforces the assumption that males outweigh women in paid work in most developing nations, including Ghana. Most respondents held the procurement officer position, with 141

representing (45%), followed by the assistant procurement officer position with 95 respondents (31%). Further, 71 out of the 307 respondents (23.1%) were in the position of storekeepers. Similarly, 145 (47.2%) survey respondents had 11-20 years of job experience in the public procurement entity. In addition, 129 respondents (42%) worked in public procurement entities for 1 to 10 years. In addition, 32 respondents had worked for 21-30 years in public entities representing (10.4%). Finally, 1 respondent indicated they have worked in the public procurement entity for 30 years and more. According to the data, pressure from stakeholders and industry actors could lead to sustainable public procurement. Public officials with 1-10 years of experience are more likely to adopt innovative procurement procedures. People with this much experience are open to change and eager to challenge customary procurement.

Again, all respondents have some level of education, with a Bachelor's degree being the most significant and a Doctorate (PhD) being the lowest rating. As a result, 135 of the 307 respondents (44%) claimed to have a Bachelor's degree. Furthermore, 93 of the respondents (30.3 %) elaimed to have a Diploma/HND degree, 74 (24.1%) claimed to have a Master' degree, and the remaining 5 claimed to have a Doctoral (Ph.D.) (1.6%). Most respondents have a bachelor's or diploma, signifying postsecondary education on average. This increases Ghana's value on the global market for sustainable development contributions through public procurement. Again, Table 4 obtained information in relation to the professional affiliations of respondents. It was revealed that, 166 (54.1%) of respondents had CIPS certification. In the same vein, 82(26.7%) of respondents had certification in CILT, 16(5.2%) also had certification in GIPS, while 40(13%) of the respondents claimed to have other professional certifications as well. Further, ICAG and ACCA professional certificates recorded the least responses, with 2(.7%) and 1(.3), respectively. This indicates that survey respondents were public procurement experts directly involved in buying goods, works, and services for their organizations.

Finally, respondents were asked to indicate which sector their procurement entity belonged to: ministries, departments, and agencies (MDAs), metropolitan, municipal and district assemblies (MMDAs), state-owned enterprises, hospitals and tertiary institutions. From table 4 above, it was revealed that respondents working in the MDAs recorded the least number of respondents with 16(5.2%) responses and MMDAs recorded the highest number of responses 119(38.8%). Further, 82 (26.7%) of responses out of the 307 indicated that, they worked in the various hospitals, and 69(22.5%) worked in the state-owned enterprises while 21(6.8%) worked in the tertiary institutions. Metropolitan, municipal, and district assembly received the most responses because it provides government services to the country's residents by purchasing goods, works, and services, such as road construction and social amenities.

4.3 Measurement Model Assessment

The partial least squares structural equation modelling (PLS-SEM) estimator was employed to model the effect of isomorphic pressures on sustainable public procurement adoption and performance. The software employed in processing and analysing the data used in this study was Version 4.0 of Smart-PLS. The usage of PLS-SEM requires two investigations: the investigation of the measurement model and the structural model. The measurement model assessment ensures that all the criteria are valid and reliable. Also, the structural model assessment helps test the hypothesis (explaining the relationship between variables). The results were generated using 5,000 consistent bootstrap samples, and the hypotheses were tested under a 5% significance level.

4.4 Reflective measurement Model Evaluation

A reflective measurement model evaluates the nexuses among the indicators and their measurement models (constructs) and between the latent constructs (Hair, Hult, Ringle, & Sarstedt, 2017). Thus, the indicators of a reflective model are wellconnected to each construct via loadings and indicate bivariate relationships between the indicator and the construct. The reflective measurement model aimed to ensure the construct measures' reliability and validity to authenticate the appropriateness of their presence in the path model (Hair, Sarstedt, Hopkins, & Kuppelwieser, 2014). The criteria employed to evaluate the psychometric properties of the model are the factor loadings, indicator and construct reliability, convergent and discriminant validity, and collinearity.

4.5 Indicator Factor Loadings

Indicator factor loadings illustrate how each indicator item effectively describes the main construct within the study's context. Analysis of the indicator loadings of a reflectively measured latent construct is the first stage of determining the indicator's quality in the study. Hair, Sarstedt, Hopkins, and Kuppelwieser (2014) recommended that factor loadings be at least 0.70 and higher. Similarly, Hair, Hult, Ringle, Sarstedt, Danks, and Ray (2021) suggested a minimum desired threshold of 0.708 for an indicator factor loading. Therefore, for an item loading to be considered an acceptable measure of the quality of its latent construct, it must be higher than the 0.708 threshold with a corresponding T-value greater than 1.96. The final model presented in Figure 2 and 3 shows that all the indicators are valid.



Figure 3:T-Values

4.6 Reliability and Convergent Validity

Internal consistency reliability was assessed after the factor loadings. An indicator is reliable if it has the potency to be replicated in different contexts and is certain for a similar or invariably the same outcome. The reliability of an indicator is measured using Cronbach Alpha (CA), Rho_A, and the composite reliability (CR). In order to ensure that there is no issue with construct reliability, composite reliability must be greater or equal to 0.70. Similarly, Cronbach alpha and Rho_A values must be greater than 0.70 to ensure that the constructs are valid.

Similarly, convergent validity is the extent to which a measure correlates positively with alternative measures of the same construct (Hair et al., 2016). Hair *et al.* (2022) proposed a minimum extracted average variance (AVE) of 0.5 to demonstrate convergent validity. An AVE value of "0.50" or greater indicates that "the definition represents more than half the variance of its indicators". Lastly, the collinearity procedure involves examining the variance inflation factors to ensure no bias in the regression estimate. It tests whether correlations among constructs are not substantial. Latent variable scores of the endogenous variables are used to calculate variance inflated factors (VIF) values. The study applied the recommended threshold of <3.3 (Kock, 2015). The findings of the reliability and convergent validity are presented in Table 3.

	Loadings	Inner	Cronbach's		Composite	Average
		VIF	alpha	Rho_A	reliability	variance
						extracted
						(AVE)
Coercive			0.910	0.920	0.928	0.649
COE1	0.737	1.873				
COE2	0.747	2.019				
COE3	0.861	3.011				
COE4	0.768	2.589				
COE5	0.863	3.518				
COE6	0.836	2.734				
COE7	0.814	2.591				
Mimetic			0.872	0.879	0.907	0.663
MIM1	0.827	2.474				
MIM2	0.812	2.118				
MIM3	0.758	1.634				
MIM4	0.881	2.843				
MIM5	0.787	1.714				
Normative			0.897	0.904	0.919	0.618
N1	0.816	2.679				
N2	0.757	2 2 5 8				
N3	0.836	2.530				
N4	0.827	3 519				
N5	0.784	3 673				
N6	0.750	2 202				
N7	0.730	2.202				
Sustainable	0.724	2.122	0.028	0 032	0 040	0.634
nerformance			0.720	0.952	0.740	0.034
SP1	0 790	2 5/10				
SP10	0.726	2.547 2.178				
SD11	0.720	2.170				
SD1	0.821	2.074				
SF 2 SD 2	0.030	2 0 9 0				
SF3 SD4	0.011	5.060				
5P4 SD5	0.807	2.312				
SP3	0.797	2.301				
SP/	0.786	2.959				
SP9	0.763	2.620	0.007	0.007	0.010	0.64.6
Sustainable			0.896	0.896	0.918	0.616
procurement adoption	0.722	1.000				
SPPAI	0.732	1.986				
SPPA2	0.787	2.216				
SPPA3	0.796	2.319				
SPPA4	0.812	2.521				
SPPA5	0.772	2.282				
SPPA6	0.835	2.865				
SPPA7	0.753	2.273				

Table 3:Reliability and convergent validity

Source: Field survey (2023)

The findings presented in Table 1 show that all the constructs are valid and reliable. The Cronbach Alpha, composite reliability, and the average variance extracted (AVE) values are greater than the recommended threshold of 0.70 for Cronbach Alpha and composite reliability. The Cronbach Alpha values ranged from 0.872 to 0.928, while Composite reliability (CR) ranged from 0.907 to 0.940, all of which fell between the threshold of 0.70 or more proposed by Hair et al., (2014). Similarly, the AVE values are greater than the recommended threshold of 0.50, and all the Outer VIFs are less than 3.3. The findings imply that the constructs are valid, reliable, and free from common method bias.

4.7 Discriminant Validity

Discriminant validity is the extent to which a construct is truly distinct from other constructs by empirical standards. Thus, establishing discriminant validity implies that a construct is unique and captures phenomena not represented by other constructs in the model (Hair et al., 2016). The discriminant validity was primarily tested using the Heterotrait-Monotrait (HTMT) ratio. It is noted that, the HTMT ratio is regarded as a better and quality measure of discriminant validity as compared to Fornell and Larcker's (1981) criterion (Hair et al., 2014), thus recommended for testing discriminant validity by Sarstedt et al. (2014). Hair *et al.* (2020) recommend a threshold of 0.90 for HTMT correlation values. The correlation between and across individual constructs must be less than 0.90, particularly when the constructs are similar (Henseler *et al.*, 2015). The result presented in Table 4 shows that all HTMT correlation values are within the 0.90 range. This implies that there is no discriminant validity issue in this study.

	1	3	3	4	5
Coercive (1)					
Mimetic (2)	0.805				
Normative (3)	0.825	0.796			
Sustainable performance (4)	0.845	0.899	0.839		
Sustainable procurement adoption (5)	0.794	0.770	0.746	0.808	

Table 4:HTMT Correlation

Source: Field survey (2023)

4.5 Assessment of Structural Model

When all criteria to assess the measurement model have been satisfied and results meet the required thresholds, the next step is to evaluate the structural model. The structural model shows the hypothesized relationship between constructs. Before the significance of the structural relationships is assessed, some fit indices were first examined. According to Hair et al. (2018), the structural model was assessed for collinearity issues. This was followed by an examination of the model's predictive power through the coefficient of determination, R^2 and the effect size (f^2). All endogenous constructs were assessed for collinearity in the structural model.

4.6 Multicollinearity Statistics

As posited by Hair et al. (2014), "collinearity diagnostic is first examined to ensure that the path coefficients are free from bias and reduce significant levels of collinearity among the predictor constructs" The outcomes of the VIF from Table 5 show that the paths are devoid of multicollinearity with maximum VIF of 2.651, which is below the threshold of 3.3 as suggested by Kock, (2015).

	Sustainable	Sustainable procurement
	performance	adoption
Coercive		2.649
Mimetic		2.353
Normative		2.651
Sustainable performance		
Sustainable procurement	1.00	
adoption		
Source: Field survey (2023)		

Table 5:Collinearity statistics

Source: Field survey (2023)

4.7 R-Squared

The coefficient of determination (R^2) measures a model's predictive accuracy and joint significance. It represents the variance in the endogenous construct explained by all the exogenous constructs. R^2 values range from 0 to 1, where higher values indicate high predictive accuracy. Since R^2 values increase with a number of predictors, adjusted R^2 is recommended because it controls for complexity in the model and is useful when comparing models. The findings show that the R2 values for sustainable performance and sustainable procurement adoption are 0.559 and 0.615, respectively. 61.5% of sustainable procurement adoption variance was explained jointly by coercive, mimetic, and normative pressures. Similarly, sustainable public procurement adoption explained 55.9% of the variance in sustainable performance.

	R-square	R-square adjusted
Sustainable performance	0.559	0.558
Sustainable procurement adoption	0.615	0.611
Source: Field survey (2023)		

Table 6:R-Squared

4.8 F-Square

F-square (f^2) is the change in R^2 when an exogenous variable is removed from the model. With the threshold of Small (0.0 < effect size < 0.15); Medium (0.15 < effect size < 0.35); Large (effect size > 0.35). The findings indicates that removing the exogenous variable of coercive, mimetic, and normative pressure will have a small or weak effect on the R^2 value for the endogenous value of sustainable public procurement adoption (0.148), (0.078) and (0.044) respectively. Similarly, removing the exogenous variable of sustainable public procurement adoption from the model will substantially or largely affect the R^2 value for the endogenous value of sustainable performance (1.268).

	Sustainable	Sustainable procurement
	performance	adoption
Coercive		0.148
Mimetic		0.078
Normative		0.044
Sustainable performance		
Sustainable procurement	1.268	
adoption		

Table 7:F-square

Source: Field survey (2023)

4.9 Hypothesis Testing

The individual research hypothesis was examined after determining whether or not the measurement model satisfies the PLS-SEM criteria. The hypotheses were examined by looking at the direction and strength of the relationship using the path coefficient. The significance level was determined using t-statistics produced from 5000 consistent bootstraps, a 2-tailed test suggested by Hair *et al.* (2014). According to Hair *et al.* (2014), the t-statistics must be greater than 1.96, and the p-values must be lower than 0.05 for the hypothesis to be statistically significant. Table 8 presents the findings obtained from applying the PLS-SEM test to the hypothesis in light of the research objectives.

Hypothesis	β	М	T stat	P values	Decision
H1a: Coercive -> Sustainable	0.389	0.388	5.564	0.000	Accepted
procurement adoption					
H1b: Normative -> Sustainable	0.211	0.213	3.134	0.002	Accepted
procurement adoption					
H1c: Mimetic -> Sustainable	0.266	0.266	5.650	0.000	Accepted
procurement adoption					
H2: Sustainable procurement adoption	0.748	0.749	30.254	0.000	Accepted
-> Sustainable performance					
		4			
Specific indirect effect					
H3a: Coercive -> Sustainable	0.291	0.291	5.374	0.000	Accepted
procurement adoption -> Sustainable					
performance					
H3b: Normative -> Sustainable	0.158	0.160	3.198	0.001	Accepted
procurement adoption -> Sustainable					
performance					
H3c: Mimetic -> Sustainable	0.199	0.200	5.091	0.000	Accepted
procurement adoption -> Sustainable					
performance					

Table 8:Hypothesis testing

Source: Field survey (2023)

Objective 1: To examine the relationship between isomorphic pressures and sustainable procurement adoption in Ghana.

The first objective examined the relationship between isomorphic pressures (coercive, normative and mimetic) on sustainable procurement adoption. The study's findings revealed that isomorphic pressures positively and significantly affected sustainable procurement adoption. Particularly, the hypotheses indicated that sustainable public procurement adoption is influenced by coercive pressures (*H1a*) ($\beta = 0.389$, t= 5.564, p = 0.000), normative pressure (*H1b*) ($\beta = 0.211$, t=3.134, p = 0.002) and mimetic pressure (*H1c*) ($\beta = 0.266$, t= 5.650, p = 0.000). All three hypotheses were supported. The findings imply that a percentage change in coercive, normative and mimetic pressure will enhance sustainable procurement adoption by 38.9%, 21.1% and 26.6%, respectively. A careful analysis of the findings indicates that coercive pressure affects sustainable procurement adoption more than normative and mimetic pressure. The findings conclude that; isomorphic pressures positively enhance the adoption of sustainable procurement in the public sector of Ghana.

Objective 2: To examine the effect of sustainable procurement adoption on the sustainable performance of public procurement entities in Ghana

The second objective examined the effect of sustainable procurement adoption on sustainable performance. The study's findings revealed a positive and significant effect between sustainable procurement adoption and sustainable performance with (β = 0.748, t= 30.254, p = 0.000). The study's findings imply that; a unit change in sustainable procurement adoption will enhance sustainable performance by 74.8%. The findings mean that when public entities adopt and practice sustainable procurement effectively, it will enhance sustainable performance.
Objective 3: To assess the mediating role of sustainable procurement adoption in the relationship between isomorphic pressures and sustainable performance in Ghana

The third objective assessed the mediating effect of sustainable procurement adoption on the relationship between isomorphic pressures and sustainable performance. The mediating effect was assessed using the specific indirect effect. The specific indirect effect of sustainable procurement adoption was assessed on three pressures (coercive, normative, and mimetic) and sustainable performance. The findings revealed a positive and significant mediating effect of sustainable procurement adoption on isomorphic pressures and sustainable performance. The specific findings revealed that sustainable procurement adoption mediates the relationship between coercive ($\beta = 0.291$, t= 5.374, p = 0.000), normative ($\beta = 0.158$, t= 3.198, p = 0.001), and mimetic ($\beta = 0.199$, t= 5.091, p = 0.000) on sustainable performance. The direct and specific indirect effects between the variables were positive and significant, indicating a partial mediation. The findings conclude that sustainable procurement adoption partially mediates the relationship between isomorphic pressures and sustainable performance.

4.10 Discussion of findings

The study generally sought to examine the influence of isomorphic pressures (coercive, normative, and mimetic) on sustainable public procurement adoption and performance in Ghana. Three main objectives guided the general objective of the study. This section of the chapter discusses the objectives of the study.

Effect of Isomorphic Pressures on Sustainable Public Procurement Adoption in Ghana.

The discussion presented below is based on the study's analysis and results on the effect of isomorphic pressures on sustainable public procurement adoption. From this objective, three hypotheses were formulated to test the isomorphic pressures on sustainable public procurement adoption. These hypotheses were:

H1a: Coercive pressures have a significant positive influence on sustainable public procurement adoption in Ghana.

The results and analysis from the survey confirm the hypothesized statement that coercive pressures have a significant positive influence on sustainable public procurement adoption in Ghana. One possible reason could be that Ghana, as a developing country, has a definite framework for SPP, which promotes sustainable public procurement in Ghana. According to Adjei-Bamfo et al., (2019), governments of developed economies enact rules and regulations using coercive pressure more effectively and as a result, most public procurement entities of developing countries have had some coercion being exerted on suppliers and customers by regulatory bodies like the Public Procurement Authority PPA. Public procurement entities believe that once non-compliance to coercive pressures warrants operational sanctions or legal suits against them, such entities rationalise their operations to conform to the desired standards to avoid being alleged to be operationally wayward, which mainly comes with some regulatory sanctions. Hence, entities tend to comply with coercive pressures by harnessing sustainable public procurement to attain social, economic and environmental legitimacy. The findings of the study on coercive pressures having a significant positive influence on sustainable public procurement adoption are in line with results from empirical studies undertaken by some authors (Cao et al.,2014; Zhang et al., 2016; Mosocha & Fatoki, 2018; Raj et al.,2020; Latif et al.,2020; Amoako, 2021; Acquah et al.,2021) as identified in the literature review. The studies mentioned above collectively and unanimously asserted and confirmed that coercive pressures of the isomorphic pressures significantly and positively influence green practices adoption, sustainable public procurement adoption, and green procurement adoption. Therefore, the study's findings affirm hypothesis H1a as coercive pressures significantly positively influence sustainable public procurement adoption in Ghana.

H1b: There is a significant positive influence of normative pressures on sustainable public procurement adoption in Ghana.

The study's findings confirmed that normative pressures significantly and positively affect sustainable public procurement adoption in Ghana. Ghana's organizations can adopt sustainable procurement as a normative expectation by adhering to international standards, complying with government policies and regulations, and addressing the concerns of civil society and the general public. In the long run, this can help the nation achieve its sustainable development goals by enhancing its reputation, enhancing social and environmental outcomes, and increasing access to global markets. The findings from H1b aligns with a number of scholarly work Cao et al., (2014); Zhang et.al., (2016); Al-Balush and Tuni, (2018); Amoako, (2021) and Acquah et al., (2021) as all findings from empirical evidence prove that normative pressures of the isomorphic pressures positively and significantly influence sustainable procurement adoption. Yank and Kang's (2020) findings revealed that normative pressure has a strong positive relationship with environmental management

practices. Similarly, Acquah et al. (2020) findings revealed that normative pressures robustly influence the adoption of green procurement practices. That is to say, normative pressure positively influences green procurement practices.

H1c: Mimetic pressures have a significant positive influence on sustainable public procurement adoption in Ghana.

The findings of the study found mimetic pressure to have a positive significant impact on sustainable procurement adoption. The reason for the findings can be attributed to the fact that sustainable procurement practices are a complex phenomenon where multiple agents operate simultaneously, and as result, entities in developing countries like Ghana mimic the good practices of their counterparts from developed economies in order to build a good reputation in the world market (Guler et al., 2002; Keulemans & Van de Walle, 2017). Ghana's organizations may be motivated to adopt sustainable procurement practises through various factors, including observing successful peers, recognizing sustainability as a potential competitive advantage, mitigating risks, utilizing professional networks, and responding to supplier influence.

This phenomenon has the potential to result in enhanced environmental and social outcomes, increased competitiveness, and alignment with global sustainability trends, thereby significantly contributing to achieving Ghana's sustainable development goals. The findings of H1c corroborate with the findings of other empirical studies in literature (Cao et al., 2014; Dubey et al., 2017; Mosocha & Fatoki, 2018; Yang & Kang, 2019; Abdul et al., 2019; Latif et al., 2020; Raj et al., 2020; Ajibike et al., 2020; Amoako, 2021; Acquah et al., 2021). Findings from all these scholars proved convincing that mimetic pressures of isomorphic pressures influence the adoption of green and sustainable practices. The findings of the study support and affirms H1c,

thus, there is a significant positive influence of mimetic pressures on sustainable public procurement adoption in Ghana.

H2: Effect of sustainable public procurement adoption on sustainable

performance in Ghana

The second objective sought to examine the effect of sustainable procurement adoption on sustainable performance in Ghana. One hypothesis was formulated and tested at the end of the analysis, proving that sustainable public procurement adoption significantly influences sustainable performance. The findings imply that when public procurement entities adopt sustainable procurement, entities are invariably bound to obtain a desirable outcome of sustainable performance. Like all other 193 united nations member countries, Ghana is trying to prioritise sustainable outcomes as a critical end product of every item purchased or procured. The findings align with the study of Grob, and Benn, (2014); Adjei-Bamfo, (2017); Roman, (2017); Mani et al., (2018); Giacomo et al., (2019); Gouda and Saranga, (2020); Pudjijono et al., (2022).

The findings show that adopting sustainable public procurement tends to increase Ghana's sustainable performance index. Implementing sustainability results in various organizational benefits such as improved environmental outcomes, cost savings, increased consumer surplus, positive social benefits, and so on (Brammer & Walker, 2011; Roman, 2017). Authors like Giacomo et al. (2019) argued that sustainable procurement practices initiatives could lead to better economic, environmental, and social sustainability. This is because sustainable purchasing may seem more expensive than traditional purchasing. Still, it could be cheaper in the long run because of lower costs for maintenance and disposal. Gouda and Saranga (2020) also posited that activities related to sustainable public procurement help the performance of sustainability. Using "green purchasing" in China has improved the

performance of organizations (Zhu & Sarkis, 2007). Similarly, Mani et al. (2018) show that when social sustainability is adopted in public procurement, both the buyer's social reputation and the supplier's social performance improve. The study affirms and supports the hypotheses 2 (H₂).

H3: The mediating role of sustainable procurement adoption in the relationship between isomorphic pressures and sustainable performance.

The last and final objective guiding the study was to assess the mediating role of sustainable procurement adoption in the relationship between isomorphic pressures (coercive, normative, mimetic) and sustainable performance. The study's findings revealed that sustainable procurement adoption partially mediates the relationship between isomorphic pressures and sustainable performance. The presence of sustainable procurement adoption serves as a mediator, indicating that organisations implementing sustainable procurement practises are more inclined to demonstrate enhanced sustainable performance (Song et al., 2017). Organisations can aptly address isomorphic pressures by incorporating sustainability factors into procurement procedures, resulting in improved environmental, social, and economic results. For example, organisations that adhere to regulatory mandates by implementing sustainable procurement practises are more inclined to distribute and advance societal well-being. Likewise, organisations that conform to industry standards and adhere to best practises in sustainable procurement are more inclined to attain industry recognition and enhance their sustainable performance (Ghosh, 2019).

In conclusion, organisations that effectively address institutional pressures by implementing sustainable procurement practises have the potential to improve their legitimacy and engage stakeholders, thereby positively influencing their sustainable performance. Organisations can enhance their sustainable performance outcomes by implementing sustainable procurement practises, which enable them to effectively address regulatory requirements, industry norms, and institutional pressures (Khodaparastiet al., 2020). The findings of the study align with the study of Acquah et al. (2023); Acquah et al. (2021); Zhu and Sarkis (2006); Song et al. (2017); Blome et al. (2014); Grob and Benn (2014). The study affirms and supports the hypotheses 3 (H₃).

4.11 Chapter Summary

The chapter mainly focused on the survey and analysis of the data gathered. Firstly, describe the demographic characteristics of respondents, including their gender, academic qualifications, years of experience in public entities position, a professional affiliation of respondents, and sectors in which respondents belong. And finally, the study statistical findings of direct and specific indirect relationships were analysed with the PLS-SEM approach. The findings revealed that coercive, normative and mimetic pressures positively influenced sustainable public procurement adoption. Also, the findings revealed that, sustainable procurement adoption positively enhances sustainable performance. Lastly, a mediator variable of sustainable procurement adoption partially mediated the relationship between isomorphic pressures (coercive, normative and mimetic pressures) and sustainable performance.

CHAPTER FIVE

SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

5.0 Introduction

This chapter of the study consists of the presentation of key findings, conclusions and some recommendations made based on the conclusions drawn from the study. The conclusion from the various hypothesis tested was discussed in this chapter. Based on the study's limitations and conclusions, recommendations were suggested for future research.

5.1 Summary

The study's goal was to examine the impact of isomorphic pressures on sustainable public procurement adoption and performance in Ghana. This study was prompted by the need to inculcate sustainability in procurement of commodities, works and services to improve sustainable development. The increasing concerns of environmental, economic and social impacts of public procurements in contracts and purchases of public organisations have prompted the need to inculcate sustainability in the purchasing and contracting activities of public organisations. From this point of view, sustainable procurement will only be useful and efficient if it can lead to changes in how public organizations usually buy and contract for goods, works, and services. According to the study's objectives, 3 hypotheses were tested to accomplish the study's objectives. The hypotheses, derived from three objectives of the study were used to guide the direction of the literature review of the study in a systematic order. The literature review section of the study was divided into three parts; theoretical, the empirical investigation and a review of concepts which depicts the pictorial or diagrammatical view of the various concepts and constructs of the study. The review of literature was carried out, using a theory base approach, adopting the institutional theory and the stakeholder theory which aided in the empirical links between the various variable of interest of the study; coercive pressures, normative pressures, mimetic pressures, sustainable public procurement adoption and sustainable performance.

Quantitative based approach was employed in the research, with the positivism philosophical paradigm backing the quantitative justification of study. An explanatory design was employed to explain the relationships and causal effects of the exogenous variables and endogenous variables chosen for the study. The survey was then conducted in Ghanaian public procurement institutions using a convenience sampling technique to sample respondents for the study. The study questioned 307 public procurement entities, including Metropolitan, Municipal, and District Assemblies, Ministries, Departments, and Agencies, State Owned Enterprises, Hospitals, and Tertiary Institutions. According to the Public Procurement Authority, there are 660 public procurement entities in Ghana (PPA).

In addition, a closed-ended questionnaire was distributed and collected from study participants. Data for the study were gathered using measurement scales that have been validated and proven to be reliable. The study used the Statistical Package for Social Sciences (SPSS version 26) to analyze and collate data. The process data was analysed using the Partial Least Squares-Structural Equation Modelling (PLS-SEM).

5.2 Summary of Key Findings

The Study analysed three objectives. The first objective examined the relationship between isomorphic pressures and sustainable procurement adoption in Ghana. The study analysed the individual items (thus normative, coercive and mimetic pressures) on sustainable procurement adoption in Ghana. The findings of the study

University of Education, Winneba http://ir.uew.edu.gh

proved that isomorphic pressures (normative, coercive and mimetic pressures) have a positive and significant effect on sustainable procurement adoption in Ghana. All the three hypotheses were supported.

The second objective sought to examine the effect of sustainable procurement adoption on sustainable performance in Ghana. One hypothesis was formulated and tested at the end of the analysis, proving that sustainable public procurement adoption significantly influences sustainable performance.

The third objective assessed the mediating role of sustainable procurement adoption in the relationship between isomorphic pressures (coercive, normative, mimetic) and sustainable performance. The study's findings revealed that sustainable procurement adoption partially mediates the relationship between isomorphic pressures and sustainable performance. The presence of sustainable procurement adoption serves as a mediator, indicating that organisations implementing sustainable procurement practises are more inclined to demonstrate enhanced sustainable performance.

5.3 Conclusion

From the findings of the study, it was conclusively established that isomorphic (coercive, mimetic and normative) pressures accounts for a positive significant influence in sustainable public procurement adoption. Again, the study proved that sustainable procurement adoption enhances sustainable performance. Lastly, the findings proved that sustainable procurement adoption partially mediates the relationship between isomorphic factors and sustainable performance

5.4 Recommendations

Based on the findings of the study, the following recommendations are made to public procurement entities, government and the public procurement authority, Ghana in order to improve sustainable public procurement and performance in Ghana.

University of Education, Winneba http://ir.uew.edu.gh

It is recommended that; institutional regulatory bodies should spearhead the sustainable operations of public procurement. By cohesion and giving stringent rules and regulations governing the procurement of goods, works and service, suppliers and contractors alike who would bid for public procurements would be very much aware of sustainable criteria in the selection process and as such will adhere to all regulations regarding sustainability in public procurement. Despite institutionalised regulatory bodies spearheading the sustainable operations of public entities, by exercising their legitimate responsibilities, the study recommends that management of regulatory bodies such as the Public Procurement Authority (PPA), Environmental Protection Agency (EPA), Ghana Standards Authority (GSA), Public Utilities and Regulatory Commission (PURC), should continue to positively engage public entities to undertake procurements while promoting sustainable manufacturing operations.

Thus, these regulatory bodies should incentivise manufacturer firms (suppliers or contractors) by involving or granting them a representative during stakeholder meetings and conferences to promote sustainability across the business environment. Lastly, to ensure such regulatory institutions live up to the core mandate, the government and other key stakeholders should actively allocate adequate resources (financial, human and statutory) for the effective and efficient management of such regulatory organisations since their vibrancy status depends on how adequately they are catered for by their sponsors' such government and other international regulatory institutions.

Further, it is recommended that the public entities should continue to comply with external pressures in mimetic pressures. The adherence to these pressures enables them to operate under sustainable parameters. Again, public entities should ensure that sustainable outcomes are stringently ascribed as core elements of their strategic business objectives. Also, the public procurement authority in collaboration with the government and public entities, should establish a harmonious working relationship with other procurement authorities from other countries. By harmonising the operational interactions among such organisations will help foster positive standards of sustainable public procurement adoption and performance in the procurement system in Ghana. Through these operational interactions, the authority would invest more into productive resource offerings like specialised trainings for employees, workshops on sustainability in procurement as well as the need for change from a traditional procurement system into a more sustainable one.

5.5 Contribution to Knowledge

The study's findings represent a significant and valuable contribution to existing body of literature. The research contributed to the existing body of knowledge by establishing the relationship between isomorphic pressures and sustainable performance. Similarly, the research findings supported the premise that sustainable procurement adoption mediates the relationship between isomorphic pressure on sustainable performance.

5.6 Suggestions for Future Studies

Future research was suggested based on the study's results and limitations.; firstly, the survey adopted a cross sectional survey, which can statistically limit findings of a study and hence it is suggested for future comparative studies to adopt a more longitudinal survey. Again, the study's model focused the isomorphic pressure and its influence on sustainable public procurement adoption in Ghana, ignoring the antecedents of these pressures. Understanding these antecedents can drive public procurement sustainability, hence future studies are advised to look at the antecedents of the isomorphic pressures and its influence on sustainability in procurement. Finally, the role of other possible mediator variables can be employed in future studies to explore the influence of isomorphic pressures through (these mediator variable) on sustainable procurement adoption.



REFERENCES

- Abu-Rumman, A. (2021). Transformational leadership and human capital within the disruptive business environment of academia. World Journal on Educational Technology: Current Issues,13(2), 178-187. https://doi.org/10.18844/ wjet.v13i2.5652
- Acquah, I. S. K., Naude, M. J., & Sendra-García, J. (2021). Supply chain collaboration in the petroleum sector of an emerging economy: Comparing results from symmetrical and asymmetrical approaches. *Technological Forecasting and Social Change*, 166, 120568. https://doi.org/10.1016/j.techfore.2020.120568
- Acquah, I.S.K., Essel, D., Baah, C., Agyabeng-Mensah, Y. and Afum, E. (2021), "Investigating the efficacy of isomorphic pressures on the adoption of green manufacturing practices and its influence on organizational legitimacy and financial performance", *Journal of Manufacturing Technology Management*, (32)7, 1399-1420. https://doi.org/10.1108/JMTM-10-2020-0404
- Adèr, H. J., Adèr, H. J., & Mellenbergh, G. J. (2008). The main analysis phase. Advising on research methods: A consultant's companion, 357-386.
- Adjei, A. B. (2010). Sustainable public procurement: a new approach to good governance. *Seul: IPPC4*.
- Adjei-Bamfo, P., Maloreh-Nyamekye, T., & Ahenkan, A. (2019). The role of egovernment in sustainable public procurement in developing countries: A systematic literature review. *Resources, Conservation and Recycling*,142, 189-203. https://doi.org/10.1016/j.resconrec.2018.12.001
- Adusei, C., & Awunyo-Vitor, D. (2015). Implementation challenges of the public procurement act by selected metropolitan, municipal and district assemblies in the Ashanti Region, Ghana. *iBusiness*, 7(01), 39. https://doi.org/10.4236 /ib.2015.71005
- Agyekum, A.K., Fugar, F.D.K., Agyekum, K., Akomea-Frimpong, I. and Pittri, H. (2022), "Barriers to stakeholder engagement in sustainable procurement of public works", *Engineering, Construction and Architectural Management*, https://doi.org/10.1108/ECAM-08-2021-0746
- Agyepong, A. O., & Nhamo, G. (2017). Green procurement in South Africa: perspectives on legislative provisions in metropolitan municipalities. *Environment, development and sustainability*, 19(6), 2457-2474. https://doi. org/10.1007/s10668-016-9865-9
- Ahari, M. N., Azad, A., Alizadeh-Zarei, M., Ebadi, A., Parand, A., & Mohammadi, P. (2018). Development and validity of the school interim competency of performance skill battery scale (SICPSBS). *International Journal of Pediatrics-Mashhad*, 6(11), 8451-8473. https://doi.org/10.22038/ijp.2018.32460.2860

- Ajibike, W. A., Adeleke, A. Q., Mohamad, F., Nawi, M. N. M., Bamgbade, J. A., Riazi, S. R. M., & Ahmad, M. F. (2020). Achieving environmental sustainability in malaysian construction industry through institutional pressure. *Journal of Critical Reviews*, 7(7), 1159-1167.http://dx.doi.org/10.31838/jcr.07.07.212
- Amenta, E. (2005). State-centered and political institutional theory: Retrospect and prospect. *The handbook of political sociology*, 96-114. https://doi.org/10 .1017/cbo9780511818059.006
- Ameyaw, C., Mensah, S., & Osei-Tutu, E. (2012). Public procurement in Ghana: the implementation challenges to the public procurement law 2003 (Act 663). *International Journal of Construction supply chain management*,2(2), 55-65. https://doi.org/10.14424/ijcscm201012-55-65
- Amoako, G. K., Adam, A. M., Arthur, C. L., & Tackie, G. (2021). Institutional isomorphism, environmental management accounting and environmental accountability: a review. *Environment, Development and Sustainability*, 23(8), 11201-11216.https://doi.org/10.1007/s10668-020-01140-y
- Ankrah, G. (2016). Public procurement amendment bill 2015 undergoes second reading.
- Aragón-Correa, J. A., Hurtado-Torres, N., Sharma, S., & García-Morales, V. J. (2008). Environmental strategy and performance in small firms: A resource-based perspective. *Journal of environmental management*, 86(1), 88-103. https://doi.org/10.1016/j.jenvman.2006.11.022
- Asare, D., 2022. Is Ghana's Public Procurement System Hurting or Saving the Public Purse?, Centre for Education Policy Analysis. United States of America. Retrieved from https://policycommons.net/artifacts/2333037/is-ghanas-publicprocurement-system-hurting-or-saving-the-public-purse/3093700/ on 30 Oct 2022. CID: 20.500.12592/871710.
- Asare, E. N., & Prempeh, K. B. (2016). Measures of ensuring value for money in public procurement: a case of selected polytechnics in Ghana. *Journal of Logistics Management*, 5(1), 22-31.
- Asthana, A. (2020). *Multicollinearity* [PowerPoint slide] Department of statistics, University of Lucknow, Lucknow.
- Avella, J. R. (2016). Delphi panels: Research design, procedures, advantages, and challenges. *International Journal of Doctoral Studies*, 11, 305.
- Baah, C., Opoku-Agyeman, D., Acquah, I. S. K., Agyabeng-Mensah, Y., Afum, E., Faibil, D., & Abdoulaye, F. A. M. (2021). Examining the correlations between stakeholder pressures, green production practices, firm reputation, environmental and financial performance: evidence from manufacturing SMEs. *Sustainable Production and Consumption*, 27, 100-114. https://doi.org/ 10.1016/j.spc.2020.10.015

- Badi, S., & Murtagh, N. (2019). Green supply chain management in construction: A systematic literature review and future research agenda. *Journal of cleaner* production,223, 312-322. https://doi.org/10.1016/j.jclepro.2019.03.132
- Bagozzi, R. P., Yi, Y., & Phillips, L. W. (1991). Assessing construct validity in organizational research. *Administrative science quarterly*, 421-458. https://doi.org/10.2307/2393203
- Bai, C., Kusi-Sarpong, S., & Sarkis, J. (2017). An implementation path for green information technology systems in the Ghanaian mining industry. *Journal of Cleaner Production*,164, 1105-1123. https://doi.org/10.1016/ j.jclepro. 2017. 05.151
- Berkowitz, R., & Benbenishty, R. (2012). Perceptions of teachers' support, safety, and absence from school because of fear among victims, bullies, and bully-victims. *American journal of orthopsychiatry*, 82(1), 67. https://doi.org/10.1111/j.1939-0025.2011.01132.x
- Berrone, P., Fosfuri, A., Gelabert, L., & Gomez-Mejia, L. R. (2013). Necessity as the mother of 'green' inventions: Institutional pressures and environmental innovations. *Strategic Management Journal*, 34(8), 891-909. https://doi.org/ 10.1002/smj.2041
- Bhakoo, V., & Choi, T. (2013). The iron cage exposed: Institutional pressures and heterogeneity across the healthcare supply chain. *Journal of Operations Management*, 31(6), 432-449. https://doi.org/10.1016/j.jom.2013.07.016
- Boomsma, M. J. (2008). Sustainable procurement from developing countries. *Royal Tropical Institute, Amsterdam.*
- Braam, G. J., de Weerd, L. U., Hauck, M., & Huijbregts, M. A. (2016). Determinants of corporate environmental reporting: The importance of environmental performance and assurance. *Journal of cleaner production*, *129*, 724-734. https://doi.org/10.1016/j.jclepro.2016.03.039
- Brammer, S. and Walker, H. (2011), "Sustainable procurement in the public sector: an international comparative study", *International Journal of Operations & Production Management*, 31(4), 452-476. https://doi.org/10.1108/ 014435 71111119551
- Campbell, D. T., & Fiske, D. W. (1959). Convergent and discriminant validation by the multitrait-multimethod matrix. *Psychological bulletin*, 56(2), 81. https://doi.org/ 10.1037/h0046016
- Campbell, J. L. (2007). Why would corporations behave in socially responsible ways? An institutional theory of corporate social responsibility. Academy of management Review, 32(3), 946-967. https://doi.org/10.5465/ amr.2007. 25275684

- Cao, H., Chang, R., Kallal, J., Manalo, G., McCord, J., Shaw, J. and Starner, H. (2014), "Adaptable apparel: a sustainable design solution for excess apparel consumption problem", *Journal of Fashion Marketing and Management*, 18 (1), 52-69. https://doi.org/10.1108/JFMM-08-2012-0046
- Carter, C. R., & Jennings, M. M. (2004). The role of purchasing in corporate social responsibility: a structural equation analysis. *Journal of business Logistics*, 25(1), 145-186.https://doi.org/10.1002/j.2158-1592.2004.tb00173.x
- Castka, P., & Balzarova, M. A. (2008). ISO 26000 and supply chains—On the diffusion of the social responsibility standard. *International journal of production economics*, *111*(2), 274-286.https://doi.org/10.1016/j.ijpe.2006.10.017
- Chen, A.J.W., Boudreau, M. and Watson, R.T. (2008), "Information systems and ecological sustainability", *Journal of Systems and Information Technology*, 10(3), 186-201.https://doi.org/10.1108/13287260810916907
- Cheng, W., Appolloni, A., D'Amato, A., & Zhu, Q. (2018). Green Public Procurement, missing concepts and future trends–A critical review. *Journal of Cleaner Production*, 176, 770-784.https://doi.org/10.1016/j.jclepro.2017.12.027
- Chikwere, G. U., Simon, S. K., Dzandu, S. S. K., & Dza, M. (2019). Compliance issues with public procurement regulations in Ghana. *International Journal of Business and Management*, 14(5), 1-8. https://doi.org/10.5539/ijbm.v14n5p1
- Chin, W. W. (2010). How to write up and report PLS analyses. In *Handbook of partial least squares* (pp. 655-690). Springer, Berlin, Heidelberg. https://doi.org/10. 1007/978-3-540-32827-8 29
- Chowdhury, S., Dey, P. K., Rodríguez-Espíndola, O., Parkes, G., Tuyet, N. T. A., Long, D. D., & Ha, T. P. (2022). Impact of Organisational Factors on the Circular Economy Practices and Sustainable Performance of Small and Medium-sized Enterprises in Vietnam. *Journal of Business Research*, 147, 362-378. https://doi.org/10.1016/j.jbusres.2022.03.077
- Clemens, E. S., & Cook, J. M. (1999). Politics and institutionalism: Explaining durability and change. *Annual review of sociology*, 441-466. <u>https://doi.org/10.1146/annurev.soc.25.1.441</u>
- Colwell, S. R., & Joshi, A. W. (2013). Corporate ecological responsiveness: Antecedent effects of institutional pressure and top management commitment and their impact on organizational performance. *Business Strategy and the Environment*, 22(2), 73-91. https://doi.org/10.1002/bse.732
- Craney, T. A., & Surles, J. G. (2002). Model-dependent variance inflation factor cutoff values. *Quality engineering*, 14(3), 391-403.https://doi.org/10.1081/QEN-120001878
- Creswell, J. W. (2014). A concise introduction to mixed methods research. SAGE publications.
- Creswell, J. W., & Clark, V. L. P. (2017). *Designing and conducting mixed methods research*. Sage publications.

- Creswell, J. W., & Creswell, J. D. (2017). *Research design: Qualitative, quantitative, and mixed methods approaches*. Sage publications.
- Crotty, M. J. (1998). The foundations of social research: Meaning and perspective in the research process. *The foundations of social research*, 1-256.
- Daddi, T., Testa, F., Frey, M., & Iraldo, F. (2016). Exploring the link between institutional pressures and environmental management systems effectiveness: An empirical study. *Journal of environmental management*, 183, 647-656. https://doi.org/10.1016/j.jenvman.2016.09.025
- Dai, J., Xie, L., & Chu, Z. (2021). Developing sustainable supply chain management: The interplay of institutional pressures and sustainability capabilities. *Sustainable Production and Consumption*, 28, 254-268.

https://doi.org/10.1016/j.spc.2021.04.017

- De Giacomo, M. R., & Bleischwitz, R. (2020). Business models for environmental sustainability: Contemporary shortcomings and some perspectives. *Business Strategy and the Environment*, 29(8), 3352-3369. https://doi.org/10. 1002/bse.2576
- De Giacomo, M. R., Testa, F., Iraldo, F., & Formentini, M. (2019). Does green public procurement lead to life cycle costing (LCC) adoption? *Journal of Purchasing* and Supply Management, 25(3), 100500. https://doi.org/10.1016 /j.pursup.2018.05.001
- Delmas, M., Russo, M. V., & Montes-Sancho, M. J. (2007). Deregulation and environmental differentiation in the electric utility industry. *Strategic Management Journal*, 28(2), 189-209. https://doi.org/10.1002/smj.578
- DiMaggio, P. J., & Powell, W. W. (1983). The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American sociological review*, 147-160.https://doi.org/10.2307/2095101
- DiMaggio, P. J., & Powell, W. W. (2002). Institutional isomorphism and collective rationality in organizational fields. *Strategy: Critical Perspectives on Business and Management*, 1(1983), 283.
- Dubey, R., Gunasekaran, A., Papadopoulos, T., Childe, S. J., Shibin, K. T., & Wamba, S. F. (2017). Sustainable supply chain management: framework and further research directions. *Journal of cleaner production*, 142, 1119-1130. https://doi.org/10.1016/j.jclepro.2016.03.117
- Dza, M., Gapp, R., & Fisher, R. (2015). Taking the professionalism out of the profession: a study of procurement and Africa. *International Journal of Procurement Management*,8(3), 251-271.
- Edman, Å. (2021). Central purchasing bodies in Sweden. In *Centralising Public Procurement* (pp. 297-315). Edward Elgar Publishing.

- Esfahbodi, A., Zhang, Y., Watson, G., & Zhang, T. (2017). Governance pressures and performance outcomes of sustainable supply chain management–An empirical analysis of UK manufacturing industry. *Journal of cleaner production*, *155*, 66-78.https://doi.org/10.1016/j.jclepro.2016.07.098
- Essel, E. A. (2021). The Consequences of Public Procurement and Its Associated Irregularities in Ghana. *Academic and Applied Research in Military and Public Management Science*, 20(1), 55-65.
- Etse, D., McMurray, A., & Muenjohn, N. (2022). The effect of regulation on sustainable procurement: Organisational leadership and culture as mediators. *Journal of Business Ethics*, 177(2), 305-325. https://doi.org/10.1007/s10551-021-04752-0
- Foo, P. Y., Lee, V. H., Ooi, K. B., Tan, G. W. H., & Sohal, A. (2021). Unfolding the impact of leadership and management on sustainability performance: Green and lean practices and guanxi as the dual mediators. *Business Strategy and the Environment*, 30(8), 4136-4153.https://doi.org/10.1002/bse.2861
- Fornell, C., & Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of marketing research*, *18*(1), 39-50.https://doi.org/10.1177/002224378101800313
- Freeman, R. Edward. (1984). Strategic Management: A Stakeholder Approach (Boston: Pitman Publishing Inc.).
- Freudenreich, B., Lüdeke-Freund, F., & Schaltegger, S. (2020). A stakeholder theory perspective on business models: Value creation for sustainability. *Journal of Business Ethics*, 166(1), 3-18.https://doi.org/10.1007/s10551-019-04112-z
- Friedman, A. L., & Miles, S. (2006). Stakeholders: Theory and practice. OUP Oxford.
- Fuertes Giné, L., Vanacore, E., & Hunka, A. D. (2022). Public Procurement for the Circular Economy: Comparative Study of Sweden and Spain. *Circular Economy and Sustainability*, 1-21.https://doi.org/10.1007/s43615-022-00150-4
- Galaskiewicz, J., & Wasserman, S. (1989). Mimetic processes within an interorganizational field: An empirical test. *Administrative science quarterly*, 454-479.https://doi.org/10.2307/2393153
- Gandolfi, F., & Stone, S. (2018). Leadership, leadership styles, and servant leadership. *Journal of Management Research*, 18(4), 261-269.
- Garson, G. D. (2012). Testing statistical assumptions. Asheboro, NC: Statistical Associates Publishing.
- Geisser, S. (1974). A predictive approach to the random effect model. *Biometrika*, 61(1), 101-107.https://doi.org/10.1093/biomet/61.1.101
- Gholami, R., Sulaiman, A. B., Ramayah, T., & Molla, A. (2013). Senior managers' perception on green information systems (IS) adoption and environmental performance: Results from a field survey. *Information & management*, 50(7), 431-438.https://doi.org/10.1016/j.im.2013.01.004

- Gidigah, B.K., Agyekum, K. and Baiden, B.K. (2022), "Defining social value in the public procurement process for works", *Engineering, Construction and Architectural Management*, 29 (6), 2245-2267.https://doi.org/10.1108/ECAM-10-2020-0848
- Gong, M., Simpson, A., Koh, L., & Tan, K. H. (2018). Inside out: The interrelationships of sustainable performance metrics and its effect on business decision making: Theory and practice. *Resources, Conservation and Recycling*, 128, 155-166.

https://doi.org/10.1016/j.resconrec.2016.11.001

- Gouda, S. K., & Saranga, H. (2020). Pressure or premium: what works best where? Antecedents and outcomes of sustainable manufacturing practices. *International Journal of Production Research*, 58(23), 7201-7217. https://doi.org/10.1080/00207543.2020.1717010
- Grandia, J. (2016). Finding the missing link: Examining the mediating role of sustainable public procurement behaviour. *Journal of Cleaner Production*, *124*, 183-190. https://doi.org/10.1016/j.jclepro.2016.02.102
- Grandia, J. J., & Kruyen, P. P. (2020). Assessing the implementation of sustainable public procurement using quantitative text-analysis tools: A large-scale analysis of Belgian public procurement notices. *Journal of Purchasing and Supply Management*, 26(4), 100627.https://doi.org/10.1016/j.pursup.2020.100627
- Gray, G. C., & Silbey, S. S. (2014). Governing inside the organization: Interpreting regulation and compliance. *American Journal of Sociology*, *120*(1), 96-145. https://doi.org/10.1086/677187
- Gray, R., Kouhy, R. and Lavers, S. (1995), "Corporate social and environmental reporting: a review of the literature and a longitudinal study of UK disclosure", *Accounting, Auditing & Accountability Journal*, 8 (2) 47-77. https://doi.org/10.1108/09513579510146996
- Grob, S., & Benn, S. (2014). Conceptualising the adoption of sustainable procurement: an institutional theory perspective. *Australasian journal of environmental management*, 21(1), 11-21.https://doi.org/10.1080/14486563.2013.878259
- Guler, I., Guillén, M. F., & Macpherson, J. M. (2002). Global competition, institutions, and the diffusion of organizational practices: The international spread of ISO 9000 quality certificates. *Administrative science quarterly*, 47(2), 207-232. https://doi.org/10.2307/3094804
- Gunningham, N. (2010). Enforcement and compliance strategies. In *The Oxford* handbook of regulation. Oxford University Press. https://doi.org/10.1093/oxfordh b/9780199560219.003.0007
- Hahn, R., & Kühnen, M. (2013). Determinants of sustainability reporting: a review of results, trends, theory, and opportunities in an expanding field of research. *Journal of cleaner production*, 59, 5-21. https://doi.org/10.1016/ j.jclepro .2013.07.005

- Hair, J. F., Risher, J. J., Sarstedt, M., & Ringle, C. M. (2018). When to use and how to report the results of PLS-SEM. *European Business Review*. https://doi.org/10.1108/EBR-11-2018-0203
- Hair, J. F., Ortinau, D. J., & Harrison, D. E. (2010). *Essentials of marketing research* (2). New York, NY: McGraw-Hill/Irwin.
- Hair, J. F., Ringle, C. M., & Sarstedt, M. (2011). PLS-SEM: Indeed a silver bullet. Journal of Marketing theory and Practice, 19(2), 139-152. https://doi.org/10.2753/MTP1069-6679190202
- Hair, J.F., Hult, G.T.M., Ringle, C.M. and Sarstedt, M. (2022), A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM), Sage, Thousand Oaks, CA.
- Hami, N., Muhamad, M. R., & Ebrahim, Z. (2015). The impact of sustainable manufacturing practices and innovation performance on economic sustainability. *Procedia Cirp,26*, 190-195. https://doi.org/10.1016/j.procir.2014.07.167
- Harland, C., Telgen, J., Callender, G., Grimm, R., & Patrucco, A. (2019). Implementing government policy in supply chains: an international coproduction study of public procurement. *Journal of supply chain management*, 55(2), 6-25. https://doi.org/10.1111/jscm.12197
- Hatch, J. A. (2010). Rethinking the relationship between learning and development: Teaching for learning in early childhood classrooms. In *The Educational Forum* (Vol. 74, No. 3, pp. 258-268). Taylor & Francis Group. https://doi.org/10.1080/00131725.2010.483911
- Henseler, J. (2017). Partial Least Squares Path Modeling. In: Leeflang, P., Wieringa, J., Bijmolt, T., Pauwels, K. (eds) Advanced Methods for Modeling Markets. *International Series in Quantitative Marketing*. Springer, Cham. https://doi.org/10.1007/978-3-319-53469-5 12
- Henseler, J., Ringle, C. M., & Sarstedt, M. (2015). A new criterion for assessing discriminant validity in variance-based structural equation modelling. *Journal* of the academy of marketing science, 43(1), 115-135. https://doi.org/10.1007/ s11747-014-0403-8
- Heugens, P. P., & Lander, M. W. (2009). Structure Agency (and other quarrels): A meta-analysis of institutional theories of organization. *Academy of management journal*,52(1), 61-85. https://doi.org/10.5465/amj.2009.36461835
- Hillman, A. J., Keim, G. D., & Luce, R. A. (2001). Board composition and stakeholder performance: Do stakeholder directors make a difference? *Business & Society*,40(3), 295-314. https://doi.org/10.1177/000765030104000304
- Hsu, C., Choon Tan, K., Hanim Mohamad Zailani, S. and Jayaraman, V. (2013), "Supply chain drivers that foster the development of green initiatives in an emerging economy", *International Journal of Operations & Production Management*, 33 (6), 656-688.https://doi.org/10.1108/IJOPM-10-2011-0401

- Hughes, D. J., Lee, A., Tian, A. W., Newman, A., & Legood, A. (2018). Leadership, creativity, and innovation: A critical review and practical recommendations. *The Leadership Quarterly*,29(5), 549-569. https://doi.org/10.1016 /j.leaqua. 2018.03.001
- Hull, C. E., & Rothenberg, S. (2008). Firm performance: The interactions of corporate social performance with innovation and industry differentiation. *Strategic management journal*, 29(7), 781-789. https://doi.org/10.1002/smj.675
- Huq, F. A., & Stevenson, M. (2020). Implementing socially sustainable practices in challenging institutional contexts: Building theory from seven developing country supplier cases. *Journal of Business Ethics*, 161(2), 415-442. https://doi.org/10.1007/s10551-018-3951-x
- Ibrahim, M. J. (2017). Introductory chapter: Economics, natural resources and sustainable development. *Emerging Issues in Economics and Development*, 1-4.
- Jain, S., Singhal, S., Jain, N. K., & Bhaskar, K. (2020). Construction and demolition waste recycling: Investigating the role of theory of planned behaviour, institutional pressures and environmental consciousness. *Journal of Cleaner Production*, 263, 121405. https://doi.org/10.1016/j.jclepro.2020.121405
- Jaja, S. A., Gabriel, J. M. O., & Wobodo, C. C. (2019). Organizational isomorphism: The quest for survival. Noble International Journal of Business and Management Research, 3(5), 86-94.
- Jennings, P. D., & Zandbergen, P. A. (1995). Ecologically sustainable organizations: An institutional approach. *Academy of management review*, 20(4), 1015-1052.

https://doi.org/10.5465/amr.1995.9512280034

- Johnsen, T. E., Miemczyk, J., & Howard, M. (2017). A systematic literature review of sustainable purchasing and supply research: Theoretical perspectives and opportunities for IMP-based research. *Industrial Marketing Management*,61, 130-143.https://doi.org/10.1016/j.indmarman.2016.03.003
- Jöreskog, K. G. (1971). Statistical analysis of sets of congeneric tests. *Psychometrika*, 36(2), 109-133.https://doi.org/10.1007/BF02291393
- Kamble, S. S., Gunasekaran, A., & Gawankar, S. A. (2020). Achieving sustainable performance in a data-driven agriculture supply chain: A review for research and applications. *International Journal of Production Economics*, 219, 179-194.https://doi.org/10.1016/j.ijpe.2019.05.022
- Kauppi, K. and Hannibal, C. (2017), "Institutional pressures and sustainability assessment in supply chains", *Supply Chain Management*, 22(5), 458-472.

https://doi.org/10.1108/SCM-01-2017-0004

- Kaya, B., Abubakar, A. M., Behravesh, E., Yildiz, H., & Mert, I. S. (2020). Antecedents of innovative performance: Findings from PLS-SEM and fuzzy sets (fsQCA). Journal of Business Research, 114, 278-289. https://doi.org/10.1016/j.jbusres.2020.04.016
- Ke, W., Liu, H., Wei, K. K., Gu, J., & Chen, H. (2009). How do mediated and nonmediated power affect electronic supply chain management system adoption? The mediating effects of trust and institutional pressures. *Decision Support Systems*, 46(4), 839-851.https://doi.org/10.1016/j.dss.2008.11.008
- Ketchen Jr, D. J., & Hult, G. T. M. (2007). Bridging organization theory and supply chain management: The case of best value supply chains. *Journal of operations management*, 25(2), 573-580.https://doi.org/10.1016/j.jom.2006.05.010
- Keulemans, S. and Van de Walle, S. (2017), "Cost-effectiveness, domestic favouritism and sustainability in public procurement: A comparative study of public preferences", *International Journal of Public Sector Management*, 30(4), 328-341.https://doi.org/10.1108/IJPSM-10-2016-0169
- Kline, R. B. (2011). Convergence of structural equation modelling and multilevel modelling.
- Kock, N. (2015). Common method bias in PLS-SEM: A full collinearity assessment approach. *International Journal of e-Collaboration (ijec)*, 11(4), 1-10.

https://doi.org/10.4018/ijec.2015100101

- Kock, N., & Lynn, G. (2015). Lateral collinearity and misleading results in variancebased SEM: An illustration and recommendations. *Journal of the Association* for information Systems, 13(7).
- Kothari, A. (2008). Protected areas and people: the future of the past. *Parks*, 17(2), 23-34.
- Kothari, C. R. (2004). Research methodology: Methods and techniques. New Age International.
- Krejcie, R. V., & Morgan, D. W. (1970). Determining sample size for research activities. *Educational and psychological measurement*, 30(3), 607-610. https://doi.org/10.1177/001316447003000308
- Küçükoğlu, M. T., & Pınar, R. İ. (2016). The mediating role of green organizational culture between sustainability and green Innovation: A research in Turkish companies. *Business & Management Studies: An International Journal* 6, 64-85. https://doi.org/10.20944/preprints201611.0122.v1
- Kuhn, D. (2011). What is scientific thinking and how does it develop?.
- Kusi-Sarpong, S., Sarkis, J., & Wang, X. (2016). Assessing green supply chain practices in the Ghanaian mining industry: A framework and evaluation. *International Journal of Production Economics*, 181, 325-341. https://doi.org/10.1016/j.ijpe.2016.04.002

- Kyeremeh, E. (2022). Upholding Sustainable Public Procurement Practices in Ghana: A Health Service Perspective. American International Journal of Business Management. 5(6). 1-13.
- Latan, H., Jabbour, C. J. C., de Sousa Jabbour, A. B. L., Wamba, S. F., & Shahbaz, M. (2018). Effects of environmental strategy, environmental uncertainty and top management's commitment on corporate environmental performance: The role of environmental management accounting. *Journal of cleaner production*, 180, 297-306. https://doi.org/10.1016/j.jclepro.2018.01.106
- Leal Filho, W., Shiel, C., Paço, A., Mifsud, M., Ávila, L. V., Brandli, L. L., ... & Caeiro, S. (2019). Sustainable Development Goals and sustainability teaching at universities: Falling behind or getting ahead of the pack?. *Journal of Cleaner Production*, 232, 285-294. https://doi.org/10.1016/j.jclepro.2019.05.309
- Leedy, P. D. & Ormrod, J. E. (2010). *Practical Research: Planning and Design* (9th ed.). Boston, MA: Pearson.
- Leiter, J. (2005). Structural isomorphism in Australian non-profit organizations. Voluntas: International Journal of Voluntary and Nonprofit Organizations, 16(1), 1-31. https://doi.org/10.1007/s11266-005-3230-1
- Lenox, M. J., & Eesley, C. E. (2009). Private environmental activism and the selection and response of firm targets. *Journal of Economics & Management Strategy*, 18(1), 45-73. https://doi.org/10.1111/j.1530-9134.2009.00207.x
- Li, F., & Ding, D. Z. (2013). The effect of institutional isomorphic pressure on the internationalization of firms in an emerging economy: Evidence from China. Asia Pacific Business Review, 19(4), 506-525. https://doi.org/10.1080/13602381.2013.807602
- Liao, Z. (2018). Environmental policy instruments, environmental innovation and the reputation of enterprises. *Journal of Cleaner Production*, 171, 1111-1117.

https://doi.org/10.1016/j.jclepro.2017.10.126

- Lin, C. Y., & Ho, Y. H. (2011). Determinants of green practice adoption for logistics companies in China. *Journal of business ethics*, 98(1), 67-83. https://doi. org/10.1007/s10551-010-0535-9
- Lin, R. J., & Sheu, C. (2012). Why do firms adopt/implement green practices? an institutional theory perspective. *Procedia-Social and Behavioral Sciences*, 57, 533-540. https://doi.org/10.1016/j.sbspro.2012.09.1221
- Liu, J., Feng, Y., Zhu, Q. and Sarkis, J. (2018), "Green supply chain management and the circular economy: Reviewing theory for advancement of both fields", *International Journal of Physical Distribution & Logistics Management*, 48(8), 794-817. https://doi.org/10.1108/IJPDLM-01-2017-0049
- Liu, X., Liu, B., Shishime, T., Yu, Q., Bi, J., & Fujitsuka, T. (2010). An empirical study on the driving mechanism of proactive corporate environmental management in China. *Journal of environmental management*, 91(8), 1707-1717. https://doi.org/10.1016/j.jenvman.2010.03.011

- Lowry, P. B., & Gaskin, J. (2014). Partial least squares (PLS) structural equation modelling (SEM) for building and testing behavioural causal theory: When to choose it and how to use it. *IEEE transactions on professional communication*, 57(2), 123-146. https://doi.org/10.1109/TPC.2014.2312452
- Malhotra, N., & Birks, D. F. (2007). An applied approach. *Marketing research*. *London: Prentice Hall*.
- Malhotra, N., Nunan, D., & Birks, D. (2017). Marketing research: An applied approach. Pearson.
- Mani, V., Jabbour, C. J. C., & Mani, K. T. (2020). Supply chain social sustainability in small and medium manufacturing enterprises and firms' performance: Empirical evidence from an emerging Asian economy. *International Journal of Production Economics*, 227, 107656. https://doi.org/10.1016/j. ijpe.2020. 107656
- Martínez-Ferrero, J., & García-Sánchez, I. M. (2017). Coercive, normative and mimetic isomorphism as determinants of the voluntary assurance of sustainability reports. *International Business Review*, 26(1), 102-118. https://doi.org/10. 1016/j.ibusrev.2016.05.009
- Masele, J. J. (2019). Modeling Green eBusiness Adoption among Small and Medium Tourism Enterprises in Tanzania. *The African Journal of Information Systems*, 11(3), 4.
- Masocha, R., & Fatoki, O. (2018). The impact of coercive pressures on sustainability practices of small businesses in South Africa. *Sustainability*, 10(9), 3032.

https://doi.org/10.3390/su10093032

- McCann, J. T., & Holt, R. A. (2010). Defining sustainable leadership. *International Journal of Sustainable Strategic Management*, 2(2), 204-210. https://doi.org/10.1504/ijssm.2010.032561
- Mensah, S., & Ameyaw, C. (2012). Sustainable procurement: the challenges of practice in the Ghanaian construction industry. In *West Africa Built Environment Research (WABER)* 2, 871.
- Metcalf, L., & Benn, S. (2013). Leadership for sustainability: An evolution of leadership ability. *Journal of business ethics*, 112(3), 369-384.

https://doi.org/10.1007/s10551-012-1278-6

- Meyer, J. W., & Rowan, B. (1977). Institutionalized organizations: Formal structure as myth and ceremony. *American journal of sociology*, *83*(2), 340-363. https://doi.org/10.1086/226550
- Michelsen, O., & de Boer, L. (2009). Green procurement in Norway; a survey of practices at the municipal and county level. *Journal of environmental management*, 91(1), 160-167. https://doi.org/10.1016/j.jenvman.2009.08.001

- Mizruchi, M. S., & Fein, L. C. (1999). The social construction of organizational knowledge: A study of the uses of coercive, mimetic, and normative isomorphism. *Administrative science quarterly*, 44(4), 653-683. https://doi.org/10.2307/2667051
- Mugenda, O. & Mugenda A. (2008). *Research methods: quantitative and qualitative approaches*.
- . Nyantakyi, A. (2018). Preparing Today for Prosperity Tomorrow: Using Sustainable Public Procurement As a Tool for Development in Ghana. *Pub. Cont. LJ*, 48, 377.
- Olya, H. G., & Altinay, L. (2016). Asymmetric modeling of intention to purchase tourism weather insurance and loyalty. *Journal of Business Research*, 69(8), 2791-2800. https://doi.org/10.1016/j.jbusres.2015.11.015
- Osei-Tutu, E., Badu, E. and Owusu-Manu, D. (2010), "Exploring corruption practices in public procurement of infrastructural projects in Ghana", *International Journal of Managing Projects in Business*, 3(2), 236-256. https://doi.org/10.1108/17538371011036563
- Papadimitriou, A. (2011). Reforms, leadership and quality management in Greek higher education. *Tertiary Education and Management*, 17(4), 355-372. https://doi.org/10.1080/13583883.2011.602705
- Pappas, I. O., & Woodside, A. G. (2021). Fuzzy-set Qualitative Comparative Analysis (fsQCA): Guidelines for research practice in Information Systems and marketing. *International Journal of Information Management*, 58, 102310. https://doi.org/10.1016/j.ijinfomgt.2021.102310
- Pappas, I. O., Giannakos, M. N., & Sampson, D. G. (2016). Making sense of learning analytics with a configurational approach. In proceedings of the workshop on smart environments and analytics in video-based learning (SE@ VBL), LAK2016.
- Park, D. H., Lee, J., & Han, I. (2007). The effect of on-line consumer reviews on consumer purchasing intention: The moderating role of involvement. *International Journal of Electronic Commerce*, 11(4), 125–148. https://doi:10.2753/JEC1086-4415110405
- Parker, C. M., Redmond, J., & Simpson, M. (2009). A review of interventions to encourage SMEs to make environmental improvements. *Environment and planning C: Government and policy*, 27(2), 279-301. https://doi.org/10. 1068/c0859b
- Peng, X., & Zhang, R. (2022). Corporate governance, environmental sustainability performance, and normative isomorphic force of national culture. *Environmental Science and Pollution Research*, 29(22), 33443-33473. https://doi.org/10.1007/s11356-022-18603-6

- Perez-Batres, L. A., Miller, V. V., & Pisani, M. J. (2011). Institutionalizing sustainability: an empirical study of corporate registration and commitment to the United Nations global compact guidelines. *Journal of Cleaner Production*, 19(8), 843-851. https://doi.org/10.1016/j.jclepro.2010.06.003
- Peter, J. P., & Churchill Jr, G. A. (1986). Relationships among research design choices and psychometric properties of rating scales: A meta-analysis. *Journal of Marketing Research*, 23(1), 1-10. https://doi.org/10.1177/0022243786023001
- Podsakoff, P. M., MacKenzie, S. B., & Podsakoff, N. P. (2012). Sources of method bias in social science research and recommendations on how to control it. *Annual review of psychology*, 63(1), 539-569.
- Poltronieri, C. F., Ganga, G. M. D., & Gerolamo, M. C. (2019). Maturity in management system integration and its relationship with sustainable performance. *Journal of Cleaner Production*, 207, 236-247 https://doi.org/10.1016/j.jclepro.2018.09.250
- Preacher, K. J., & Hayes, A. F. (2008). Asymptotic and resampling strategies for assessing and comparing indirect effects in multiple mediator models. *Behavior research methods*, 40(3), 879-891. https://doi.org/10.3758/BRM.40.3.879
- Pudjijono, A., Hartono, D. M., Hasibuan, H. S., & Nursani, D. (2022). Sustainable public procurement: Research trends and gaps. *Indian Journal of Ecology*,49(3), 945-953. https://doi.org/10.55362/IJE/2022/3620
- Purwanto, A., & Sudargini, Y. (2021). Partial least squares structural squation modeling (PLS-SEM) analysis for social and management research: a literature review. *Journal of Industrial Engineering & Management Research*,2(4), 114-123. https://doi.org/10.7777/jiemar.v2i4.168
- Quashie, M. K. (2019). Causes and Costs of Procurement Irregularities in Ghana's District Assemblies. *International Journal of Law and Society*, 2(4), 58.
- Ragin, C. C. (2008). Measurement versus calibration: A set-theoretic approach.
- Raj, A., Agrahari, A., & Srivastava, S. K. (2020). Do pressures foster sustainable public procurement? An empirical investigation comparing developed and developing economies. *Journal of Cleaner Production*, 266, 122055. https://doi.org/10.1016/j.jclepro.2020.122055
- Ramayah, T. J. F. H., Cheah, J., Chuah, F., Ting, H., & Memon, M. A. (2018). Partial least squares structural equation modelling (PLS-SEM) using smartPLS 3.0. An updated guide and practical guide to statistical analysis.
- Ramírez, R. R., & Palos-Sánchez, P. R. (2018). Environmental firms' better attitude towards nature in the context of corporate compliance. *Sustainability*, 10(9), 3321.https://doi.org/10.3390/su10093321
- Repar, N., Jan, P., Nemecek, T., Dux, D., & Doluschitz, R. (2018). Factors affecting global versus local environmental and economic performance of dairying: A case study of Swiss mountain farms. *Sustainability*, 10(8), 2940. https://doi.org/10.3390/su10082940

- Rigdon, E. E. (2012). Rethinking partial least squares path modeling: In praise of simple methods. *Long range planning*, 45(5-6), 341-358. https://doi.org/10.1016/j.lrp.2012.09.010
- Rihoux, B., Ragin, C. C., Yamasaki, S., & Bol, D. (2009). Conclusions-The way (s) ahead. *Configurational comparative methods: Qualitative comparative analysis* (QCA) and related techniques, 167-178.
- Roldán, J. L., & Sánchez-Franco, M. J. (2012). Variance-based structural equation modeling: Guidelines for using partial least squares in information systems research. In *Research methodologies, innovations and philosophies in software* systems engineering and information systems (pp. 193-221). IGI global.

https://doi.org/10.4018/978-1-4666-0179-6.ch010

- Roldán, J. L., & Sánchez-Franco, M. J. (2012). Variance-based structural equation modelling: Guidelines for using partial least squares in information systems research. In *Research methodologies, innovations and philosophies in software* systems engineering and information systems pp. 193-221 https://doi.org/10.4018/978-1-4666-0179-6.ch010
- Roman, A. V. (2017). Institutionalizing sustainability: A structural equation model of sustainable procurement in US public agencies. *Journal of cleaner* production,143, 1048-1059. https://doi.org/10.1016/j.jclepro.2016.12.014
- Roxas, B., & Coetzer, A. (2012). Institutional environment, managerial attitudes and environmental sustainability orientation of small firms. *Journal of Business Ethics*, 111(4), 461-476. https://doi.org/10.1007/s10551-012-1211-z
- Russell, S. V., Lafferty, G., & Loudoun, R. (2008). Examining tourism operators' responses to environmental regulation: the role of regulatory perceptions and relationships. *Current Issues in Tourism*, 11(2), 126-143. https://doi.org/10.2167/cit319.0
- Ryu, Y., & Sueyoshi, T. (2021). Examining the relationship between the economic performance of technology-based small suppliers and socially sustainable procurement. *Sustainability*, 13(13), 7220. https://doi.org/10.3390/su13137220
- Saeed, A., Jun, Y., Nubuor, S. A., Priyankara, H. P. R., & Jayasuriya, M. P. F. (2018). Institutional pressures, green supply chain management practices on environmental and economic performance: A two theory view. *Sustainability*,10(5), 1517. https://doi.org/10.3390/su10051517
- Saeed, A., Ramish, A., & Yusuf, I. (2019). Mimetic Pressures and Environmental Performance: Mediating Role of Internal Environmental Management.
- Saunders, M., Lewis, P., & Thornhill, A. (2009). Research methods for business students. Pearson education.
- Saunders, M., Lewis, P., & Thornhill, A. (2012). Research methods for business students (6. utg.). *Harlow: Pearson*.

- Schneider, C. Q., & Wagemann, C. (2010). Standards of Good Practice in Qualitative Comparative Analysis (QCA) and Fuzzy-Sets, *Comparative Sociology*, 9(3), 397-418.https://doi.org/10.1163/156913210X12493538729793
- Seow, P. S., Pan, G., & Tay, J. (2014). Revisiting the determinants of students' performance in an undergraduate accountancy degree programme in Singapore. *Global Perspectives on Accounting Education*, 11(3), 1-23.
- Shubham, Charan, P., & Murty, L. S. (2018). Organizational adoption of sustainable manufacturing practices in India: integrating institutional theory and corporate environmental responsibility. *International Journal of Sustainable Development & World Ecology*, 25(1), 23-34. https://doi.org/10.1080/13504509.2016.1258373
- Siedlecki, S. L. (2020). Understanding descriptive research designs and methods. *Clinical Nurse Specialist*, 34(1), 8-12. http://doi.org/10.1097/NUR.00000000000493
- Smelt, M. (2021). Environmental responsibility and performance in small and mediumsized enterprises An institutional framework on the drivers and performance benefits of environmental management system adoption for SMEs.
- Smith, N. (2010). Uneven development: Nature, capital, and the production of space. University of Georgia Press.
- Somjai, S., & Jermsittiparsert, K. (2019). Role of pressures and green supply chain management practices in enhancing the operational efficiency of firms: evidence from Thailand. *International Journal of Supply Chain* Management,8(4), 437-445.
- Sönnichsen, S. D., & Clement, J. (2020). Review of green and sustainable public procurement: Towards circular public procurement. *Journal of cleaner* production, 245, 118901. https://doi.org/10.1016/j.jclepro.2019.118901
- Stone, M. (1974). Cross-validation and multinomial prediction. *Biometrika*, 61(3), 509-515. https://doi.org/10.1093/biomet/61.3.509
- Sullivan, G. M., & Feinn, R. (2012). Using effect size—or why the P value is not enough. *Journal of graduate medical education*, 4(3), 279-282. https://doi.org/10.4300/JGME-D-12-00156.1
- Tabachnick, B. G., Fidell, L. S., & Ullman, J. B. (2007). Using multivariate statistics, 5(4), 481-498). Boston, MA: pearson.
- Taylor, B., Sinha, G., & Ghoshal, T. (2006). Research methodology: A guide to for researchers in management and social sciences. PHI Learning Pvt. Ltd.
- Teo, H. H., Wei, K. K., & Benbasat, I. (2003). Predicting intention to adopt interorganizational linkages: An institutional perspective. *MIS quarterly*, 19-49.

https://doi.org/10.2307/30036518

- Tolbert, P. S., & Zucker, L. G. (1983). Institutional sources of change in the formal structure of organizations: The diffusion of civil service reform, 1880-1935. *Administrative science quarterly*, 22-39.https://doi.org/10.2307/2392383
- Treviño-Lozano, L. (2021). Sustainable Public Procurement and Human Rights: Barriers to Deliver on Socially Sustainable Road Infrastructure Projects in Mexico. Sustainability, 13(17), 9605. https://doi.org/10.3390/su13179605
- Tsinopoulos, C., Sousa, C. M., & Yan, J. (2018). Process innovation: open innovation and the moderating role of the motivation to achieve legitimacy. *Journal of product innovation management*, *35*(1), 27-48. https://doi.org/10.1111/jpim.12374
- Üçok Hughes, M., Upadhyaya, S. and Houston, R. (2018), "Educating future corporate managers for a sustainable world: recommendations for a paradigm shift in business education", *On the Horizon*, 26(3), 194-205. https://doi.org/10.1108/OTH-01-2018-0007
- Uttam, K., & Roos, C. L. L. (2015). Competitive dialogue procedure for sustainable public procurement. *Journal of Cleaner Production*, *86*, 403-416. https://doi.org/10.1016/j.jclepro.2014.08.031
- Visser, W., & Courtice, P. (2011). Sustainability leadership: Linking theory and practice. Available at SSRN 1947221.
- Walker, H. (2015). New development: Public procurement research at IPSERA aligning research and practice, and future trends. *Public Money & Management*,35(2), 141-144. https://doi.org/10.1080/09540962. 2015.1007710
- Walker, H., & Brammer, S. (2012). The relationship between sustainable procurement and e-procurement in the public sector. *International Journal of Production Economics*, 140(1), 256-268. https://doi.org/10.1016/j.ijpe.2012.01.008
- Walker, H., Di Sisto, L., & McBain, D. (2008). Drivers and barriers to environmental supply chain management practices: Lessons from the public and private sectors. *Journal of purchasing and supply management*, 14(1), 69-85.

https://doi.org/10.1016/j.pursup.2008.01.007

- Wang, J. and Dai, J. (2018), "Sustainable supply chain management practices and performance", *Industrial Management & Data Systems*, 118(1), 2-21. https://doi.org/10.1108/IMDS-12-2016-0540
- Warachan, B. (2011). Appropriate statistical analysis for two independent groups of Likert-type data. American University.
- Witjes, S., & Lozano, R. (2016). Towards a more Circular Economy: Proposing a framework linking sustainable public procurement and sustainable business models. *Resources, Conservation and Recycling*, 112, 37-44. https://doi.org/10.1016/j.resconrec.2016.04.015

- Woodside, A. G. (2014). Embrace• perform• model: Complexity theory, contrarian case analysis, and multiple realities. *Journal of Business Research*, 67(12), 2495-2503.https://doi.org/10.1016/j.jbusres.2014.07.006
- Xie, L., Ju, T., & Xia, B. (2021). Institutional pressures and megaproject social responsibility behaviour: a conditional process model. *Buildings*, *11*(4),140. https://doi.org/10.3390/buildings11040140
- Yammarino, F. (2013). Leadership: Past, present, and future. Journal of Leadership & Organizational Studies, 20(2), 149-155. https://doi.org/10.1177/ 1548051812471559
- Yang, M. G., & Kang, M. (2020). An integrated framework of mimetic pressures, quality and environmental management, and firm performances. *Production Planning & Control*, 31(9), 709-722. https://doi.org/10.1080 /09537287.2019.1681533
- Yates, S. (2003). Doing social science research. Sage.
- Yong, J. Y., Yusliza, M. Y., Ramayah, T., Chiappetta Jabbour, C. J., Sehnem, S., & Mani, V. (2020). Pathways towards sustainability in manufacturing organizations: Empirical evidence on the role of green human resource management. *Business Strategy and the Environment*, 29(1), 212-228.

https://doi.org/10.1002/bse.2359

- Yusliza, M. Y., Yong, J. Y., Tanveer, M. I., Ramayah, T., Faezah, J. N., & Muhammad, Z. (2020). A structural model of the impact of green intellectual capital on sustainable performance. *Journal of Cleaner Production*, 249, 119334. https://doi.org/10.1016/j.jclepro.2019.119334\
- Zaid, A. A., Jaaron, A. A., & Bon, A. T. (2018). The impact of green human resource management and green supply chain management practices on sustainable performance: An empirical study. *Journal of cleaner production*, 204, 965-979. https://doi.org/10.1016/j.jclepro.2018.09.062
- Zhao, X., Lynch Jr, J. G., & Chen, Q. (2010). Reconsidering Baron and Kenny: Myths and truths about mediation analysis. *Journal of consumer research*, 37(2), 197-206. https://doi.org/10.1086/651257
- Zhu, Q., & Sarkis, J. (2007). The moderating effects of institutional pressures on emergent green supply chain practices and performance. *International journal* of production research, 45(18-19), 4333-4355. https://doi.org/10.1080/002 07540701440345
- Zhu, Q., Sarkis, J., & Lai, K. H. (2013). Institutional-based antecedents and performance outcomes of internal and external green supply chain management practices. *Journal of Purchasing and Supply Management*, 19(2), 106-117. https://doi.org/10.1016/j.pursup.2012.12.001
- Zsidisin, G. A., Melnyk, S. A., & Ragatz, G. L. (2005). An institutional theory perspective of business continuity planning for purchasing and supply management. *International journal of production research*, *43*(16), 3401-3420.

APPENDIX

QUESTIONNAIRE UNIVERSITY OF EDUCATION, WINNEBA SCHOOL OF BUSINESS DEPARTMENT OF PROCUREMENT AND SUPPLY CHAIN MANAGEMENT

Dear Sir/Madam,

The questionnaire is to solicit information on the **"influence of isomorphic pressures on sustainable public procurement adoption and performance in Ghana"** The researcher is a student at the School of Business, Department of Procurement and Supply Chain Management, University of Education, Winneba. You are hereby invited to share your views on the issue under investigation. The responses would be used for purely academic purposes and as part of the requirement for the award of a Master's degree in Procurement and Supply Chain Management. Your confidentiality is greatly assured. Therefore, no information provided will be identifiable to you or your organisation since only aggregated data will be reported in this study. You are kindly required to answer the questions as frankly as possible since you will be contributing to knowledge. Thanks for your time and accepting to participate in the study.

SECTION A: ISORMORPHIC PRESSURES

Please indicate the degree of your agreement with the following statements by ticking or clicking ($\sqrt{}$) the appropriate number 5= Strong Agreement 1=Least Agreement

Coercive Pressures							
Statements Grouper State					4	5	
COE1	If public procurement entities commit a sustainable infraction, the consequence would include negative reports by the public procurement Authority/stock market analysts	1	2	3	4	5	
COE2	There are well-defined fines and penalties for unsustainable behaviours in public procurement	1	2	3	4	5	
COE3	Entities that do not meet regulatory standards are threatened with legal prosecution.	1	2	3	4	5	
COE4	There are negative consequences for procurement entities that do not comply with government sustainability policies.	1	2	3	4	5	
COE5	Our major suppliers believe that we should adopt sustainable practices when procuring.	1	2	3	4	5	
COE6	We are likely to lose major suppliers if we do not adopt sustainable practices when procuring	1	2	3	4	5	

COE7	Our entity's success greatly depends on our participation in sustainable supply chain management	1	2	3	4	5
Mimetic pressures					4	5
MIM 1	Public entities in other developing countries that have adopted sustainable public procurement have benefited greatly	1	2	3	4	5
MIM 2	Public entities in other developing countries that have adopted sustainable public procurement are more competitive with high sustainable indexes	1	2	3	4	5
MIM 3	Leading organizations have set examples for social sustainability in public procurement for goods, works and services	1	2	3	4	5
MIM 4	Leading organizations have set examples for environmental sustainability in public procurement for goods, works and services	1	2	3	4	5
MIM 5	Leading organisations in the public sector are known for their practices that promoted environmental preservation	1	2	3	4	5
Normative Pressures					4	5
NOR 1	Nongovernmental organizations expect our entity to implement sustainable procurement	1	2	3	4	5
NOR 2	Stakeholders may not support our entity if our entity does not implement sustainable procurement	1	2	3	4	5
NOR 3	It is expected that all public procurement entities are environmentally, socially and economically responsible	1	2	3	4	5
NOR 4	Professional associations like CIPS, CILT etc. encourages entities and professionals to become more responsible in sustainable procurement	1	2	3	4	5
NOR 5	Being environmentally, socially and economically responsible is a requirement for public procurement entities to be accepted by citizens.	1	2	3	4	5
NOR 6	We actively participate in industry, trade, or professional associations that promote environmental and social supply chain management	1	2	3	4	5
NOR 7	Significant pressure to engage in sustainability procurements is placed on us from industry and professional sources that support sustainable supply chain management	1	2	3	4	5

SECTION B: SUSTAINABLE PUBLIC PROCUREMENT ADOPTION

Please indicate the level of degree of your agreement with the following statement by ticking or clicking ($\sqrt{}$) the appropriate number 5= Strong Agreement 1=Least Agreement

	Sustainable public procurement Adoption					
Statements			2	3	4	5
SPA1	Our organisation evaluates whole of life environmental friendliness of products before procuring	1	2	3	4	5
SPA2	Our organisation visits suppliers' operational sites to ensure sweatshop labour is not used	1	2	3	4	5
SPA3	Our organisation buys from Minority and Women Based Enterprises	1	2	3	4	5
SPA4	We purchase products with biodegradable packages or containers	1	2	3	4	5
SPA5	Our organisation commits its suppliers to waste reduction goals	1	2	3	4	5
SPA6	We buy products with eco-labels like energy consumption rating	1	2	3	4	5
SPA7	We make sure that our purchases do not lead to accumulation of unnecessary things	1	2	3	4	5

SECTION C: SUSTAINABLE PERFORMANCE

Please indicate the level of degree of your agreement with the following statement by ticking or clicking ($\sqrt{}$) the appropriate number 5= Strong Agreement 1=Least Agreement

	Sustainable Performance					
Econor	nic Performance	1	2	3	4	5
SP1	The entity endeavours to reduce consumption of electricity	1	2	3	4	5
SP2	The entity's cost of operations has reduced due to continuous improvement	1	2	3	4	5
SP3	The entity's economic objectives are achieved due to sustainable procurement practices.	1	2	3	4	5
Social]	l Performance				4	5
SP4	The image and reputation of entity has improved according to awards and recognitions	1	2	3	4	5
SP5	The level of satisfaction of society with the entity's actions has improved according to complaints and acknowledgements received.	1	2	3	4	5
SP6	Contribution to local prosperity (employment of local labour, using local suppliers) has increased	1	2	3	4	5
Enviro	nmental Performance	1	2	3	4	5

SP7	There is improvement in the entity's reputation due to its environmental effort	1	2	3	4	5
SP8	The entity's environmental effort has reduced air and water pollution	1	2	3	4	5
SP9	Our entity procures only eco-labelled goods	1	2	3	4	5

SECTION D: DEMOGRAPHICS

1. Gender of respondents?

Male [] Female []

2. What is your position in the entity?

Procurement officer [] Assistant Procurement officer[] Stores keeper []

3. Please indicate the number of years of experience worked in public procurement.

1-10 Years [] 11-20 Years [] 21-30 Years [] Above 30 Years []

4. Please specify your educational qualification.

Diploma [] Bachelor's degree [] Master's degree [] Doctorate degree [] Others []

5. Indicate ($\sqrt{}$) your affiliations with any of the following professional bodies.

CIPs	CILT	GIPS	ACCA	ICAG	PMI	Other
	V			4		

6. Which sector does your Entity belong?

MDAs [] MMDAs [] State Owned Enterprises [] Hospitals [] Tertiary institutions []

Thank You!!!